THE UNEQUAL CONSEQUENCES OF JOB LOSS ACROSS COUNTRIES

Antoine Bertheau Edoardo Maria Acabbi Cristina Barcelo Andreas Gulyas Stefano Lombardi Raffaele Saggio

Working Paper 29727 http://www.nber.org/papers/w29727

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 February 2022, Revised February 2023

We are very grateful to the editor Amy Finkelstein and four anonymous reviewers for their constructive comments that greatly improved the paper. We also thank Marcus Eliason, Simon Jäger, Patrick Kline, Claus Kreiner, Thomas Lemieux, Benjamin Schoefer, and Daphné Skandalis for helpful comments and seminar participants at Paris 8, VATT Institute for Economic Research Helsinki, LIER Helsinki, IFAU Uppsala, University of Copenhagen, Le Mans University, Banco de España, Catholic University of the Sacred Heart Milan, IMF, OECD, and participants at SOLE, Search and Matching Conference, AFSE, ESPE, EEA, EALE, SAEe, and the workshop BdE-CEMFI. Antoine Bertheau acknowledges financial support from the Danish National Research Foundation (Niels Bohr Professorship) and from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation program: Grant agreement 741467 FIRMNET and data access from the CASD and Denmark Statistics. Edoardo Acabbi gratefully acknowledges financial support from the Comunidad de Madrid (Programa Excelencia para el Profesorado Universitario, convenio con Universidad Carlos III de Madrid, V Plan Regional de Investigación Científica e Innovación Tecnológica), the Spanish Ministry of Science and Innovation, project PID2020-114108GB-I00, and the Fundacion Ramon Areces. Andreas Gulyas gratefully acknowledges financial support from the German Research Foundation (DFG) through CRC TR 224 (project A3). Stefano Lombardi gratefully acknowledges financial support and data access from IFAU-Uppsala (project 157-2019). Raffaele Saggio thanks the Segretariato Ufficio Statistica at the Italian Ministry of Labour for providing the LOSAI data. The views and opinions in this paper do not necessarily reflect those of our employers or the National Bureau of Economic Research.

NBER working papers are circulated for discussion and comment purposes. They have not been peer-reviewed or been subject to the review by the NBER Board of Directors that accompanies official NBER publications.

© 2022 by Antoine Bertheau, Edoardo Maria Acabbi, Cristina Barcelo, Andreas Gulyas, Stefano Lombardi, and Raffaele Saggio. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

The Unequal Consequences of Job Loss across Countries Antoine Bertheau, Edoardo Maria Acabbi, Cristina Barcelo, Andreas Gulyas, Stefano Lombardi, and Raffaele Saggio NBER Working Paper No. 29727 February 2022, Revised February 2023 JEL No. J01,J18,J23,J31

ABSTRACT

We document the consequences of losing a job across countries using a harmonized research design applied to seven matched employer-employee datasets. Workers in Denmark and Sweden experience the lowest earnings declines following job displacement, while workers in Italy, Spain, and Portugal experience losses three times as high. French and Austrian workers face earnings losses somewhere in between. Key to these differences is that Southern European workers are less likely to find employment following displacement. Loss of employer-specific wage premiums explains a substantial portion of wage losses in all countries.

Antoine Bertheau University of Copenhagen Nørregade 10, 1165 København Copenhagen Denmark antoine.bertheau@econ.ku.dk

Edoardo Maria Acabbi Universidad Carlos III de Madrid Calle Madrid 126 Getafe 28903 Spain eacabbi@emp.uc3m.es

Cristina Barcelo Bank of Spain Alcala 48 28014 Madrid Spain barcelo@bde.es Andreas Gulyas University of Mannheim andreas.gulyas@gmail.com

Stefano Lombardi VATT Institute for Economic Research Arkadiankatu 7 00101 Helsinki Finland stefano.lombardi@vatt.fi

Raffaele Saggio Department of Economics University of British Columbia 6000 Iona Drive Vancouver, BC V6T 1L4 and NBER rsaggio@mail.ubc.ca

The Unequal Consequences of Job Loss across Countries

By ANTOINE BERTHEAU, EDOARDO MARIA ACABBI, CRISTINA BARCELÓ ANDREAS GULYAS, STEFANO LOMBARDI, AND RAFFAELE SAGGIO *

We document the consequences of losing a job across countries using a harmonized research design applied to seven matched employer-employee datasets. Workers in Denmark and Sweden experience the lowest earnings declines following job displacement, while workers in Italy, Spain, and Portugal experience losses three times as high. French and Austrian workers face earnings losses somewhere in between. Key to these differences is that Southern European workers are less likely to find employment following displacement. Loss of employer-specific wage premiums explains a substantial portion of wage losses in all countries (JEL J30, J63, J64).

Losing a job entails lasting negative consequences for a worker (Jacobson, LaLonde and Sullivan, 1993). This finding is among the most influential in labor economics because it provides a simple test of how well labor markets are functioning. More efficient labor markets reallocate workers more quickly and generate lower earnings losses after job displacement. Comparing the consequences of job loss across labor markets might therefore reveal which ones are functioning better than others and why.

^{*}Bertheau: University of Copenhagen (antoine.bertheau@econ.ku.dk); Acabbi: Universidad Carlos III de Madrid (eacabbi@emp.uc3m.es); Barceló: Banco de España (barcelo@bde.es); Gulyas: University of Mannheim (andreas.gulyas@unimannheim.de); Lombardi: VATT Institute for Economic Research Helsinki, IFAU Uppsala, IZA, Uppsala Center for Labor Studies (stefano.lombardi@vatt.fi); Saggio: University of British Columbia and NBER (rsaggio@mail.ubc.ca). We are very grateful to the editor Amy Finkelstein and four anonymous reviewers for their constructive comments that greatly improved the paper. We also thank Marcus Eliason, Simon Jäger, Patrick Kline, Claus Kreiner, Thomas Lemieux, Benjamin Schoefer, and Daphné Skandalis for helpful comments and seminar participants at Paris 8, VATT Institute for Economic Research Helsinki, LIER Helsinki, IFAU Uppsala, University of Copenhagen, Le Mans University, Banco de España, Catholic University of the Sacred Heart Milan, IMF, OECD, and participants at SOLE, Search and Matching Conference, AFSE, ESPE, EEA, EALE, SAEe, and the workshop BdE-CEMFI. Antoine Bertheau acknowledges financial support from the Danish National Research Foundation (Niels Bohr Professorship) and from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation program: Grant agreement 741467 FIRMNET and data access from the CASD and Denmark Statistics. Edoardo Acabbi gratefully acknowledges financial support from the Comunidad de Madrid (Programa Excelencia para el Profesorado Universitario, convenio con Universidad Carlos III de Madrid, V Plan Regional de Investigación Científica e Innovación Tecnológica), the Spanish Ministry of Science and Innovation, project PID2020-114108GB-I00, and the Fundacion Ramon Areces. Andreas Gulyas gratefully acknowledges financial support from the German Research Foundation (DFG) through CRC TR 224 (project A3). Stefano Lombardi gratefully acknowledges financial support and data access from IFAU-Uppsala (project 157-2019). Raffaele Saggio thanks the Segretariato Ufficio Statistica at the Italian Ministry of Labour for providing the LOSAI data. The views and opinions in this paper do not necessarily reflect those of our employers.

However, such comparisons remain challenging. Meta-analyses from existing research are often clouded by differences in the sample selection, the definition of the displacement event, and the econometric specifications. These discrepancies tend to deliver different estimates on the impact of job loss and thus complicate the interpretation of competing results reached by different studies.

This paper addresses these challenges by building a harmonized dataset that combines matched employer-employee administrative registers from almost three decades and seven countries characterized by a wide range of labor market institutions (Austria, Denmark, France, Italy, Portugal, Spain, and Sweden). Our focus is on assessing the labor market effects of job displacements, defined as the permanent loss of a longterm job due to mass layoffs or establishment shutdowns for economic reasons. By adopting a common research design, a common definition of the displacement event and identical sample selection criteria, this work provides the first comparable estimates on the labor market consequences of job displacement across countries. These harmonized data and empirical methods are then used to disentangle the sources of displaced workers' pay losses both within and between countries.

The key insight of this paper is that the labor market consequences of losing a job are vastly different across Europe. Scandinavian countries experience the lowest earnings losses: five years after job displacement, earnings are about 10% lower than their pre-displacement level. By contrast, the earnings of displaced workers from Southern Europe (Italy, Spain, and Portugal) are around 30% lower than their pre-displacement levels. Austrian workers experience earnings losses in between those of the Scandinavian and Southern European countries, while French workers experience losses more similar to those of Scandinavian workers.

Interestingly, existing evidence leads to drastically different conclusions from ours. For instance, by comparing Leombruni, Razzolini and Serti (2013) to Bennett and Ouazad (2019), one would conclude that Italian displaced workers suffer lower earnings losses than Danish ones. This highlights the importance of using a harmonized research design when conducting a cross-country analysis on the consequences of job loss.

Next, we show that a large part of the cross-country differences in earnings losses is due to different responses on the extensive margin. The probability of being nonemployed five years following displacement is 20 percentage points larger for displaced workers than for non-displaced ones in Spain, Portugal, and Italy. The same estimate is only around 5 percentage points in Sweden, Denmark, and France, while it is roughly 9 percentage points in Austria. A key explanation for our findings is that a significant fraction of displaced workers from Italy, Portugal, and Spain permanently withdraws from the labor market following the displacement event. This effect appears more pronounced among women, a result that echoes the enormous differences in female employment rates observed within Europe (e.g., Christiansen et al., 2016).

After conditioning on re-employment, we find that losses in daily wages for displaced workers are less dispersed when compared to earnings, but still relatively heterogeneous, ranging from about 4% (Denmark) to 17% (Spain) five years after displacement. We then analyze the extent to which transitions from better- to worsepaying firms contribute to displaced workers' wage losses and whether these transitions differ across nations. We find that employer-specific wage policies explain a remarkably large share of the observed wage losses for all countries. The share ranges from around 35% for Spain to almost 100% for Portugal. These results are thus in line with Schmieder, von Wachter and Heining (2022) and Gulyas and Pytka (2020), who point to the importance of changes in employers' wage premiums in driving long-term wage losses from job displacement.

The remainder of the paper is structured as follows. Section I describes the data and the empirical methods used in the main analysis. Section II presents evidence on the heterogeneous impact of job loss across countries. Section III analyzes potential factors that can account for the cross-country differences depicted in Section II. Section IV concludes.

I Harmonized Research Design

Do earnings losses due to job displacement differ across countries, and if so, by how much? Table A.1, which summarizes selected papers on job displacement, shows that this question is not readily answered by comparing existing studies. In particular, earnings loss estimates for a specific country tend to vary across papers. For example, available earnings loss estimates for France vary from 16% to 36%. The reason for these different estimates is that studies on the effects of job loss use different definitions of the displacement event, sample restrictions, control group, and time periods.

These differences in the research design cloud cross-country comparisons. For example, by comparing Leombruni, Razzolini and Serti (2013) to Bennett and Ouazad (2019), one would conclude that Danish displaced workers face higher earnings losses than Italian workers. But the use of different sample restrictions (displaced workers' employers must have at least 30 employees versus no restriction on firm size) and different definitions of the mass layoff event (plant closure versus decline in firm size by over 30%) could also be driving the differences in the estimates. The definition of the control group is another important feature that tends to differ across papers. Some studies, like Jacobson, LaLonde and Sullivan (1993) and Lachowska, Mas and Woodbury (2020), impose that control workers remain *always* employed at the same employer. Table A.2 shows that imposing this tenure restriction on control workers in our analysis can double the estimated earnings losses from job displacement.

To overcome these limitations, we build a harmonized cross-country-matched employeremployee dataset by combining high-quality administrative registers from Austria, Denmark, France, Italy, Portugal, Spain, and Sweden. Table A.10 provides the data sources for each country. In our analysis we make sure to use the same variable definitions, sampling restrictions, and research design for each country. We use the resulting dataset to study job loss events due to mass layoffs occurring between at least the 1990s and the 2010s.¹ Specifically, we adopt an event study design akin to Schmieder, von Wachter and Heining (2022) where workers displaced through a mass layoff are matched to similar workers who do not experience such an event. As detailed below, we select comparison (control) workers through propensity score matching and compute dynamic job loss effects by following workers up to five years before and after the job displacement event.

A. Sample Selection and Definition of Main Outcomes

Sample selection. To limit the influence of early retirement programs, we select workers who are at most 50 years old in the year preceding the job displacement event. We consider stable jobs by sampling workers with at least three years of tenure with their main employer in the year preceding the job displacement event. The unit of analysis for the employer is the establishment.² Moreover, to identify exogenous job separations due to mass layoffs, we further restrict our sample to workers employed in establishments with at least 50 employees at the end of the pre-displacement year. Identical sampling restrictions are applied to the control workers as described below.

Definition of main outcomes. We define earnings, deflated to 2010 EUR, as the sum of yearly labor earnings (possibly from different employers) before taxation. Labor earnings include overtime, bonuses, and severance payments when available. Wages are defined as daily earnings from the main employer and are computed as labor earnings over days worked. We do not have information on hours worked for all countries (see Table A.4). A person is defined as employed if they have any positive labor earn-

¹For Spain, data on job displacements are available from 2007 onward. Table A.4 shows that the extent of information available is comparable across countries. Country-specific details concerning the construction of the matched employer-employee dataset are reported in Appendix C.

²The main employer is the establishment at which the worker's annual earnings are largest.

ings during the year. If the person is non-employed in a given year, we impute zero earnings for that particular year as is typically done in the job displacement literature (e.g., Schmieder, von Wachter and Heining, 2022).

B. Definition of Treated and Control Workers

Treatment group. Let t^* be the year of a job displacement event. We define displaced/treated workers as those satisfying the following two conditions that seek to capture exogenous and permanent job separations: (i) workers separate from their main employer in t^* , and (ii) employment at the current establishment drops by at least 30% in t^* .³

Restriction (ii) is aimed at alleviating concerns about mischaracterizing voluntary separations as layoffs.⁴ The 30% threshold is standard in the mass layoff literature (see, e.g., Davis and Von Watcher, 2011; Flaaen, Shapiro and Sorkin, 2019) and includes plant closures. We additionally use explicit information on the reason for job separation (layoff versus voluntary resignation) whenever the information is available.⁵

Control group. A potential control worker is someone who does not *concurrently* satisfy both the conditions described above that are used to define a treated worker. To match each displaced worker to one worker selected from the pool of potential control workers, we partition the data by cells defined by calendar year, gender, and industry of the displaced workers. Within each cell, we then estimate a propensity score model via probit on the likelihood of being displaced. The model includes earnings measured in $t^* - 2$ and $t^* - 3$, age, tenure, and employer size in $t^* - 1$. We also match control and treated workers by contract type (temporary versus permanent) and full-time status, both measured at $t^* - 1$, whenever this information is available. We then apply a 1:1 nearest neighbor matching algorithm without replacement to assign one control

³To focus on permanent job separations, we drop from the analysis (i.e., from both treatment and control group) workers who are recalled by their main employer within five years from displacement. Moreover, to avoid classifying mergers or domestic outsourcing events as mass layoffs, we also exclude displacement events where more than 20% of workers jointly move to another firm.

⁴An analysis based on mass layoffs permits us to study plausibly exogenous separations but has the drawback that it selects only a subset of involuntary separations. However, we find that in general the share of workers subject to a mass layoff is relatively comparable across countries (see Table 1), suggesting that the differential representativeness of mass layoff workers across countries is not a primary concern for our results.

⁵This information is available for Spain and Italy. The main results are unaffected when we do not use the reason for job separation for these countries, and thus we only focus on mass layoffs identified from administrative data (see Figure A.4). This result further suggests that focusing only on separations generated from mass layoffs measured from administrative data delivers representative estimates on the effects of job displacement (see the previous footnote and Flaaen, Shapiro and Sorkin, 2019).

worker to each treated worker. As is standard in the literature (e.g., Schmieder, von Wachter and Heining, 2022), the chosen control workers might be employed at a mass layoff firm but cannot experience a job displacement event themselves. See Appendix B for further details.

C. Summary Statistics

Table 1 presents descriptive statistics of the matched sample along with sample sizes.⁶ For each country in the study, the matching algorithm returns treated and comparison workers with well-balanced observable characteristics. In our sample, workers are, on average, between 33 and 38 years old, and between 34% and 48% are women. Treated and control workers are employed at the same employer for an average 5 to 10 years, depending on the country. Most workers work full time (81% to 89%) and on a permanent employment contract (6% to 15% have a fixed-term employment contract).

Comparing across countries, we observe that most variables are relatively balanced. However, some differences exist, such as in length of tenure. Table 1 further shows the percentage of workers involved in a mass layoff in a given year across the countries analyzed. According to this measure, the share of workers undergoing mass layoffs and fulfilling our sample restrictions is around 2% per year for most countries in our sample. Given that the definition of mass layoffs is common across countries, the fact that treated workers represent very similar shares of our samples further supports the validity of our analysis.

⁶In France, Italy, and Spain we do not have access to the universe of administrative records but rather to a random sample of individuals with all their employment spells.

	Den	mark	Swe	eden	Ita	nly	Spa	ain	Aus	stria	Fra	nce	Port	ugal
	treated	control	treated	control	treated	control	treated	control	treated	control	treated	control	treated	control
Panel A: Data structure														
Years of job loss Universe of Data	1983- YI	-2017 ES	1994- Yl	-2016 ES	1993- NO (-2016 6.5%)	2007- NO	-2019 (4%)	1987- YI	–2018 ES	1994- NO	-2016 (8%)	1992- Yl	-2017 ES
Panel B: Worker characteristics	6													
Earnings in $t^* - 3$ (EUR Th.)	40.3 (23.9)	40.2 (23.5)	32.9 (15.9)	32.8 (15.9)	23.1 (16.4)	23.1 (15.6)	22.2 (9.2)	22.1 (8.9)	29.3 (11.3)	29.3 (11.5)	28.6 (17.0)	28.8 (18.0)	14.6 (10.9)	14.7 (11.2)
Age	33.7 (9.0)	34.2 (9.1)	36.7 (7.9)	36.7 (7.9)	37.8 (7.6)	37.7 (7.8)	38.2 (6.8)	38.0 (6.9)	38.2 (7.8)	38.1 (7.8)	37.3 (7.5)	37.5 (7.6)	35.8 (7.6)	35.8 (7.7)
Female	0.37	0.37	0.34	0.34	0.40	0.40	0.41	0.41	0.42	0.42	0.36	0.36	0.48	0.48
Tenure	5.8 (3.8)	5.8 (3.8)	7.2 (4.8)	7.2 (4.8)	4.7 (1.4)	4.7 (1.3)	6.7 (3.9)	6.6 (3.8)	7.3 (4.3)	7.3 (4.3)	6.6 (5.1)	6.6 (5.0)	10.4 (7.2)	10.4 (7.2)
Temporary contract	_	_	_	_	0.06	0.06	0.14	0.15	_	_	0.09	0.09	0.13	0.13
Full time	0.81	0.81	_	_	0.86	0.85	0.87	0.86	_	-	0.88	0.88	0.89	0.89
Panel C: Employer characterist	ics													
Industry:														
Manufacturing Services Other	0.38 0.34 0.28	0.38 0.34 0.28	0.42 0.33 0.26	0.42 0.33 0.26	0.36 0.31 0.33	0.36 0.31 0.33	0.26 0.59 0.15	0.26 0.59 0.15	$\begin{array}{c} 0.47 \\ 0.13 \\ 0.41 \end{array}$	$0.47 \\ 0.13 \\ 0.41$	0.44 0.33 0.23	0.44 0.33 0.23	0.53 0.38 0.09	0.53 0.38 0.09
Establishment size	369 (603)	343 (632)	383 (669)	379 (829)	364 (284)	357 (291)	342 (789)	361 (812)	323 (373)	310 (516)	319 (434)	341 (539)	337 (494)	326 (601)
% of workers involved in a displacement event	2.	84	1.	25	3.	41	1.	80	2.	58	0.	70	1.	88
No. Workers (th.) No. Firms (th.)	201.91 7.13	201.91 10.01	108.58 6.62	108.58 15.70	66.07 22.61	66.07 28.36	14.71 13.25	14.71 13.70	55.82 1.16	55.82 5.84	28.64 8.62	28.64 19.21	171.14 7.97	171.14 44.93

Table 1: Descriptive Statistics, Matched sample

Notes: Averaged worker and employer characteristics in the matched sample, with t^* denoting the year of job loss for the treated group. Earnings are measured in $t^* - 3$, and all other variables in $t^* - 1$. The industry groups were matched at more disaggregated country-specific level but have been re-aggregated in the table for presentation purposes. Earnings are deflated and reported in 2010 Thousand Euros. Standard deviation in parentheses. % of workers involved in a displacement event reports the average share of displaced workers in relation to the overall number of workers in a given year.

II The Consequences of Job Loss across Countries

This section documents the consequences of job loss across Europe in terms of total yearly earnings, employment, and log daily wages.

A. Event Study Model

Let *i* index a treated or matched control worker, t_i^* be the calendar year when the displacement event occurs, and *k* be the number of years since displacement. We estimate the following event study model separately for each country:

$$y_{it} = \alpha_i + \lambda_t + \sum_{k=-5}^{k=5} \gamma_k \mathbf{1}\{t = t_i^* + k\} + \sum_{k=-5}^{k=5} \theta_k \mathbf{1}\{t = t_i^* + k\} \times \text{Displaced}_i + X_{it}'\beta + r_{it},$$
(1)

where y_{it} measures our dependent variable, which are total yearly earnings, employment status, or log daily wages in year *t*; *Displaced*_i is an indicator variable equal to one for treated workers who lose their job in a displacement event; and X_{it} includes age squared. The worker fixed effects α_i control for time-invariant worker characteristics, and λ_t are calendar year fixed effects. Under the assumption of parallel trends between the treated and control units, the coefficients of interest, θ_k , capture the causal effect of job loss at event time *k*. The coefficients { θ_k } are normalized relative to θ_{-3} . Standard errors are clustered at the worker level.

B. The Unequal Consequences of Job Loss across Countries

Figure 1 shows the effects of job displacement across countries, while Table 2 reports the point estimates and standard errors observed at k = 1 and k = 5. The figure reveals substantial cross-country heterogeneity regarding the impact of job loss. Starting from labor market earnings, Panel (a) shows that despite large and persistent effects of job displacement in all countries, workers displaced in Northern European countries suffer substantially lower losses in total earnings. One year after displacement, post-displacement earnings are 20% lower compared to earnings measured in the predisplacement years. Remarkably, this effect is twice as large in Southern Europe. Five years after displacement, Northern European workers still suffer a 10% loss in total earnings compared to a 30% loss for their Southern European counterparts. Austrian workers face earnings losses somewhere in between, while French workers have earnings losses more similar to what observed in the Scandinavian countries.

Panel (b) further highlights that a large part of the cross-country differences is driven by different responses on the extensive margin. Five years after displacement, the probability of being non-employed is approximately 20 percentage points higher for displaced workers in Italy, Spain, and Portugal relative to their respective control group.⁷ Conversely, the effect on employment is much more attenuated and amounts to roughly 5 percentage points in Scandinavian countries as well as France. The employment effect in Austria is somewhere in the middle and amounts to about 10 percentage points. More generally, these estimates on employment align with an older literature that has found very heterogeneous employment dynamics across countries (see, e.g., Blanchard and Portugal, 2001; Nickell, Nunziata and Ochel, 2005).

By contrast, wage losses, which are computed only for the subset of employed workers, are less dispersed across countries relative to what is observed for earnings. Yet, we still find a fair degree of heterogeneity also regarding the impact of job loss on wages.⁸ Losses in wages are lowest in Denmark (around 4%) and largest in Spain with an effect that is about four times larger compared to what we observe in Denmark. In Table A.9 we find that, after accounting for selection on re-employment using insights from Lee (2009), the cross-country heterogeneity in wage losses appears consistent with our main results on earnings displayed in Figure 1a.

⁷The pattern of the event study coefficients for employment in Portugal looks slightly different because its underlying matched employer-employee dataset (*Quadros de Pessoal*) only provides a snapshot of the labor market in October. Given our definition of displacement event, displaced workers in this country are thus employed by the long-term employer up to October of $t^* - 1$ but are always either non-employed or employed by a different employer in t^* . Shifting the event time for Portugal by one year does not qualitatively affect our main results.

⁸Panel (c) shows a spike in the wages during the job displacement year for some countries. This happens when the drop in days worked is larger than the relative loss in earnings, which typically occurs due to extra payments received by workers upon job displacement such as severance payments or accumulated leave time (see, e.g., Lachowska, Mas and Woodbury, 2020 for a similar pattern).



Figure 1: The Effect of Job Loss across Countries

Notes: The figure shows event study estimates of the job loss effects from equation (1). Estimates are relative to $t^* - 3$, where t^* is the job loss year. The coefficients in Panel (a) are re-scaled using average pre-displacement labor earnings. The outcome in Panel (b) is an indicator equal to one if a worker has at least one day of work in the corresponding year. Point estimates and standard errors are displayed in Table 2.

Figure A.1 shows that Italy is the only country with a remarkable time trend in the job loss effects, while Figure A.2 shows that, when estimating the effects of job displacement separately by men and women, the remarkable differences in earnings losses across countries remain unaltered.⁹ Finally, using the estimates from Schmieder, von Wachter and Heining (2022), Table A.6 shows that earnings losses in Germany are comparable to those observed in Austria.

All in all, when interpreting earnings losses as a proxy for how well labor markets are functioning, we obtain the clear conclusion that Northern European labor markets are more efficient in reallocating workers to new jobs, with limited earnings losses five years following the displacement event. By contrast, workers in Spain, Portugal, and Italy face significantly higher earnings losses due to displacement, which persist well after the job displacement event. Finally, wage losses, despite being less dispersed compared to earnings losses, are still quite diverse across Europe but without a clear north-south divide as the one observed for earnings.

III Understanding the Effects of Job Loss across Countries

This section addresses the following questions that arise in light of the evidence shown in Section II. First, are the differences in re-employment probabilities of displaced workers across countries driven by permanent withdrawals from the labor market or by the fact that displaced workers reallocate to very unstable jobs? Second, what is the role of employer-specific wage premiums in driving wage losses due to displacement, and is this role different across countries? Section A. provides evidence concerning the first question, while Section B. answers the last two.

A. Explaining Differences in Employment

Figure 2 displays the percentage of displaced workers who remain non-employed in a given year following the displacement event. To maximize comparability across countries, we focus on the cohort of workers displaced in the year 2010.¹⁰ The results are also presented in table format in Table A.7. Figure 2a shows remarkably different patterns across countries. Only about 20% of displaced workers in Sweden, Denmark, and France were not able to find a new job in the year right after the displacement

⁹Italian workers suffered earnings losses of around 25% in the 1990s but of 40% in the 2010s. Appendix D. shows that these larger earnings losses observed over time for Italy appears to be due to displaced workers being increasingly more likely to obtain lower-paying temporary jobs following job displacement, consistent with the findings of Woodcock (2020) for Germany.

¹⁰The results based on the full set of available mass layoff years are qualitatively similar.

	Earnin	gs	Employ	yment	Log daily wages		
Denmark							
k = 1	-17.85	(0.155)	-6.07	(0.074)	-0.07	(0.002)	
k = 5	-11.52	(0.196)	-3.59	(0.092)	-0.04	(0.002)	
Observations (th.)	4,295		4,295		3,701		
Sweden							
k = 1	-19.61	(0.240)	-8.30	(0.104)	-0.16	(0.003)	
k = 5	-11.40	(0.329)	-4.12	(0.116)	-0.08	(0.003)	
Observations (th.)	2,247		2,247		2,169		
Austria							
k = 1	-39.00	(0.247)	-17.42	(0.186)	-0.15	(0.002)	
k = 5	-21.96	(0.297)	-9.11	(0.217)	-0.12	(0.002)	
Observations (th.)	1,192		1,192		1,119		
France							
k = 1	-19.56	(0.431)	-10.68	(0.235)	-0.06	(0.003)	
k = 5	-12.25	(0.626)	-5.70	(0.320)	-0.07	(0.004)	
Observations (th.)	577		577		542		
Italy							
k = 1	-39.58	(0.386)	-27.95	(0.223)	-0.06	(0.002)	
k = 5	-27.86	(0.507)	-17.63	(0.274)	-0.07	(0.003)	
Observations (th.)	1,526		1,526		1,304		
Spain							
k = 1	-45.71	(0.551)	-28.40	(0.429)	-0.13	(0.005)	
k = 5	-32.56	(0.672)	-15.55	(0.514)	-0.17	(0.006)	
Observations (th.)	302		302		277		
Portugal							
k = 1	-35.45	(0.286)	-29.15	(0.156)	-0.03	(0.002)	
k = 5	-24.45	(0.332)	-17.32	(0.176)	-0.05	(0.002)	
<i>Observations (th.)</i>	3,491	. ,	3,491		2,692	. ,	

Table 2: Job Loss Effects on Earnings, Employment, and Wages

Notes: The table reports the coefficients θ_1 and θ_5 from the event study model (1) for each country. *k* denotes the years since displacement. The point estimates on earnings are re-scaled by the average earnings measured in the pre-displacement years, and the coefficients on earnings and employment are multiplied by 100. Standard errors are reported in parentheses and are clustered at the individual level. The number of person-year observations is in thousands.

event. This fraction is much higher, between 30% and 40% for workers in Italy and Spain, and is even higher in Portugal. Over time, these differences do not converge. The figure highlights that between 15% and 25% of workers in Spain, Italy, and Portugal never re-entered the labor market five years post displacement.

Interestingly, we detect a similar pattern while computing the hazard rates for the sample of displaced workers who *eventually* found a job within five years from displacement (see Figure 2b). For instance, while in France only 5% of returning displaced workers did not enter the labor market within two years from displacement, in Portugal this fraction is almost 30%. More generally, according to Figure 2b, the non-employment duration among displaced workers who eventually returned to the labor market remains significantly longer among Southern European displaced workers.

To quantify the overall importance of permanent displacement from the labor market in driving our effects on employment, we estimate equation (1) using as an outcome a dummy equal to one if worker *i* is not employed in year *t* and remains nonemployed in all subsequent periods (Figure A.8). Withdrawals appear to explain a significant fraction of the overall employment effects (e.g., they account for about 75%–80% of the employment effect observed for Italy, Spain, and Portugal). Table A.8 additionally shows that permanently displaced workers are systematically more likely to be women and that this gender gap is particularly large in Southern European countries.¹¹

In conclusion, the evidence presented here shows that a key explanation for our effects on employment is that displaced workers from Southern European countries have a greater chance of being *permanently* displaced from the labor market following displacement. This effect appears to be more pronounced among women. Even after conditioning on re-employment, workers from Italy, Portugal, and Spain still experience significantly longer non-employment durations following displacement.

Differences in observed characteristics, employment coverage, and institutions. Appendix B. shows that the cross-country differences in observed characteristics do not explain much of the heterogeneous effects of job displacement depicted in Figure 1a. In Appendix D., we also show that differences in coverage of self-employment or public jobs across countries are unlikely to account for the bulk of the different employment responses displayed in Figure 1b. Finally, additional analyses on the potential role of institutions are reported in the working paper version of this article (see Bertheau et al., 2022).

¹¹For instance, in Italy the permanently displaced workers are 10 percentage points more likely to be women, whereas this number is around 2-3 percentage points in Sweden and Denmark.



Figure 2: Share of Displaced Workers Who Remain Non-Employed

(b) Conditioning on Re-Employment

Notes: In Panel (a), we take the set of workers who suffered a displacement event in 2010 and plot the fraction of them who remain non-employed (i.e., have not entered the labor market yet) in each of the five years that followed the job displacement event. Panel (b) reports the same fraction among those displaced workers who eventually find a job within five years from the displacement event. Table A.7 shows the numbers reported in each panel.

B. The Role of Employer-Specific Wage Premiums

We now focus on the extent to which transitions from better- to worse-paying firms contribute to displaced workers' wage losses and whether these transitions differ across countries. Two recent studies indicate large cross-country differences in workers' ability to find similarly well-paying firms after job displacement. Lachowska, Mas and Woodbury (2020) show that displaced workers in Washington State during the Great Recession did not face a particularly significant loss of employer-specific wage premiums. In contrast, Schmieder, von Wachter and Heining (2022) show that in Germany many displaced workers move to worse-paying firms, which explains a large fraction of their wage losses.

Despite Schmieder, von Wachter and Heining (2022) partly reconciling the different results found for Germany and Washington State, the fact that these two studies use somewhat different sample restrictions and econometric specifications, as illustrated in Table A.1, makes it hard to draw firm conclusions on the importance of employer quality in explaining wage losses. Does the ability of displaced workers to find similarly well-paying jobs significantly differ across countries? To answer this question, we exploit our harmonized research design to perform a cross-country comparison of the fraction of wage losses that can be attributed to transitions to worse-paying firms after job loss.

B..1 Calculating Employer Fixed Effects

For each country, we focus on the sample of workers with positive earnings and estimate an Abowd-Kramarz-Margolis (AKM) model (Abowd, Kramarz and Margolis, 1999) for log daily wage as follows:

$$y_{it} = \alpha_i + \psi_{J(i,t)} + \lambda_t + X'_{it}\beta + u_{it},$$
⁽²⁾

where J(i,t) is the main employer of worker *i* in year *t*, α_i and $\psi_{J(i,t)}$ are worker and establishment fixed effects, λ_t are year indicators to adjust for macroeconomic conditions, and X_{it} is a cubic polynomial in age. To alleviate the concern that job loss might directly contribute to the estimates of the establishment effect in the AKM model, when estimating equation (2) we exclude treated and control workers defined in Section B.. Our focus is on the estimates of $\psi_{J(i,t)}$, which captures the time-invariant wage policy component for a given employer, which we denote as the employer-specific wage premium. As the employer-specific wage premiums correlate with productivity, they can be interpreted as capturing rents accrued by the worker from the current job (Card, Cardoso and Kline, 2016).

After estimating the AKM model, we first re-estimate the event study model (1) by using $\hat{\psi}_{J(i,t)}$ as an outcome. The interaction terms in the event study model return the change in the employer-specific wage premiums for displaced workers relative to their matched control workers, *k* years following displacement. Next, following Lachowska, Mas and Woodbury (2020), we take the ratio of the job displacement effect on the employer-specific wage premium relative to the overall job displacement effect on log wages. This gives a measure of the share of wage losses explained by changes in employer-specific wage premiums.¹²

B..2 Job Displacement Effects Due to Loss of Employer-Specific Wage Premiums

Table 3 shows the estimated loss of employer-specific wage premiums (Column 1), the total job loss effects on wages (Column 2), and the resulting share explained by employer-specific wage premiums (Column 3). The results highlight that the loss of employer-specific wage premiums is very important in explaining overall wage losses across all countries. Five years after displacement, the change in employer-specific premiums explains between 35% and 60% of wage losses in Austria, Denmark, Italy, Spain, and Sweden. In France this share is almost 70%, and in Portugal it reaches 100%. Table A.5 additionally shows that changes in employer-specific wage premiums matter in explaining the cyclicality of job loss effects.

Overall, these results suggest that the transition of displaced workers from betterto worse-paying employers is an important factor in explaining the wage losses due to displacement observed within each country. This result echoes the seminal work of Jacobson, LaLonde and Sullivan (1993), who hypothesize that a potential factor behind permanent wage losses following displacement is the loss of firm-specific rents that displaced workers accumulated with their employer before experiencing a layoff. Table 3 thus confirms this original conjecture and provides a quantitative assessment on the importance of firm-specific rents in driving the wage losses of displaced workers across very diverse labor markets. The importance of firms in driving wage losses of displaced workers highlighted by Table 3 is consistent with recent evidence that has shown that firm wage policies represent an important feature of today's labor markets for several countries (Card et al., 2018; Song et al., 2019).

¹²To compute the share of wage losses explained by losses in employer-specific pay premiums, we re-estimate the effect of job loss on log wages within the subsample of person-year observations where the associated employer belongs to the so-called largest connected set associated to equation 2 (see, e.g., Card, Heining and Kline, 2013).

	AKM er wage pi	mployer remium 1)	Log Dai (2	ly Wage 2)	Ratio (3)
Denmark					
k = 1 k = 5 <i>Observations (th.)</i>	-0.025 -0.018 3,674	(0.001) (0.001)	-0.063 -0.040 3,674	(0.002) (0.002)	$\begin{array}{c} 0.40\\ 0.44\end{array}$
Sweden					
k = 1 k = 5 <i>Observations (th.)</i>	-0.027 -0.026 1,937	(0.001) (0.001)	-0.098 -0.051 1,937	(0.003) (0.003)	0.28 0.51
Austria					
k = 1 k = 5 <i>Observations (th.)</i>	-0.061 -0.064 1,048	(0.001) (0.001)	-0.105 -0.112 1,048	(0.002) (0.002)	0.58 0.57
France					
k = 1 k = 5 <i>Observations (th.)</i>	-0.025 -0.030 489	(0.002) (0.002)	-0.036 -0.044 489	(0.003) (0.004)	0.70 0.68
Italy					
k = 1 k = 5 <i>Observations (th.)</i>	-0.023 -0.028 1,262	(0.001) (0.002)	-0.053 -0.057 1,262	(0.002) (0.003)	0.43 0.49
Spain					
k = 1 k = 5 <i>Observations (th.)</i>	-0.023 -0.045 259	(0.003) (0.004)	-0.097 -0.129 259	(0.004) (0.006)	0.24 0.35
Portugal					
k = 1 k = 5 <i>Observations (th.)</i>	-0.029 -0.044 2,525	(0.001) (0.001)	-0.029 -0.043 2,525	(0.002) (0.002)	1.00 1.01

Table 3: Loss of Employer-Specific Wage Premiums

Notes: The table reports estimates from the event study model (1), with k denoting the time since the job displacement event. Column 1 reports results where the AKM employer fixed effects is used as the dependent variable. Column 2 reports results where the log daily wage is used as the dependent variable. The resulting share of losses in log daily wages due to losses in employer-specific wage premiums is shown in Column 3. Effects on log daily wages are calculated with the sub-sample of displaced (and matched control) workers whose employer at time t belongs to the within-country largest connected set of firms associated with equation (2). Standard errors, clustered at the individual level, are reported in parentheses.

IV Conclusion

Using a harmonized research design applied to matched employer-employee datasets, we document striking differences of the impact of job loss across seven European countries. While earnings losses five years after job displacement are around 10% in Northern European countries, they are almost 30% in Southern European countries, with Austrian workers facing losses in between, and French workers' losses being more similar to those of Northern European workers. Crucially, these earnings differences appear to be driven by differences in re-employment probabilities since a significant fraction of displaced workers from Italy, Portugal, and Spain are unable to re-enter the labor market post displacement. This effect is more pronounced among women.

Focusing on wages, a key factor in driving wage losses following job displacement is reallocation to worse-paying employers. Specifically, the share of wage losses explained by losses in AKM employer-specific wage premiums ranges from 35% for Spain to basically 100% for Portugal. This result thus enriches a recent but still inconclusive literature that has analyzed the role of employer-specific wage policies in driving the wage losses following displacement (Lachowska, Mas and Woodbury, 2020; Schmieder, von Wachter and Heining, 2022; Gulyas and Pytka, 2020).

What can these results tell us about welfare? While an analysis on earnings losses is not equivalent to an analysis on income or consumption losses (e.g., Dobkin et al., 2018; Fadlon and Nielsen, 2021), we note that the countries where the earnings losses due to displacement tend to be the lowest (Denmark, Sweden) are also those where the welfare state tends to be the most generous. Conversely, the largest earnings losses are observed in countries (Italy, Spain, Portugal) where the generosity of the welfare state tends to be the lowest (Boeri, 2011). This suggests that our ranking of countries in terms of earnings losses might be preserved when also looking at income or consumption losses. However, this is clearly only speculative, and a rigorous analysis on the effects of job loss on comparable measures of consumption and income across countries represents an interesting avenue for future research.

All in all, the vastly different earnings trajectories following a job loss documented in this paper should be informative for policy makers and academics alike. Our results reveal that labor markets appear to function better in some countries than others. European policy makers should thus focus on policies that could reduce these differences, which appear even more striking when focusing on particular groups of workers, like women.

References

- **Abowd, John, Francis Kramarz, and David Margolis.** 1999. "High wage workers and high wage firms." *Econometrica*, 67(2): 251–333.
- **Bennett, Patrick, and Amine Ouazad.** 2019. "Job displacement, unemployment, and crime: Evidence from danish microdata and reforms." *Journal of the European Economic Association*.
- Bertheau, Antoine, Edoardo Maria Acabbi, Cristina Barcelo, Andreas Gulyas, Stefano Lombardi, and Raffaele Saggio. 2022. "The unequal cost of job loss across countries." National Bureau of Economic Research.
- **Blanchard, Olivier, and Pedro Portugal.** 2001. "What hides behind an unemployment rate: Comparing Portuguese and US labor markets." *American Economic Review*, 91(1): 187–207.
- **Boeri, Tito.** 2011. "Institutional Reforms and Dualism in European Labor Markets." *Handbook of Labor Economics*, 4: 1173–1236.
- **Card, David, Ana Rute Cardoso, and Patrick Kline.** 2016. "Bargaining, sorting, and the gender wage gap: Quantifying the impact of firms on the relative pay of women." *The Quarterly Journal of Economics*, 131(2): 633–686.
- **Card, David, Ana Rute Cardoso, Jörg Heining, and Patrick Kline.** 2018. "Firms and labor market inequality: Evidence and some theory." *Journal of Labor Economics*, 36(S1): S13–S70.
- **Card, David, Jörg Heining, and Patrick Kline.** 2013. "Workplace Heterogeneity and the Rise of West German Wage Inequality." *The Quarterly journal of economics*, 128(3): 967–1015.
- Christiansen, Lone Engbo, Ms Huidan Huidan Lin, Ms Joana Pereira, Petia Topalova, and Rima Turk. 2016. Unlocking female employment potential in Europe: Drivers and Benefits. International Monetary Fund.
- DADS-Etablissement. 2015. "Insee [Producteur], Base Tous Salariés : fichier Etablissements, version 2015 [Fichiers de données], Centre d'Accès Sécurisé aux Données (CASD) [Diffuseur], DOI: 10.34724/CASD.34.2498.V1."
- DADS-Panel. 2018. "Insee [Producteur], Panel tous salariés, version 2018 [Fichiers de données], Centre d'Accès Sécurisé aux Données (CASD) [Diffuseur], DOI: 10.34724/CASD.85.3955.V1."

- DADS-Poste. 2018. "Insee [Producteur], Base Tous Salariés : fichier Postes, version 2018 [Fichiers de données], Centre d'Accès Sécurisé aux Données (CASD) [Diffuseur], DOI: 10.34724/CASD.21.3867.V1."
- **Davis, Steven, and Till Von Watcher.** 2011. "Recessions and the Costs of job Loss." *Brookings Papers on Economic Activity: Fall 2011,* 1.
- Dobkin, Carlos, Amy Finkelstein, Raymond Kluender, and Matthew J Notowidigdo. 2018. "The economic consequences of hospital admissions." *American Economic Review*, 108(2): 308–52.
- Fadlon, Itzik, and Torben Heien Nielsen. 2021. "Family Labor Supply Responses to Severe Health Shocks: Evidence from Danish Administrative Records." American Economic Journal: Applied Economics, 13(3): 1–30.
- Flaaen, Aaron, Matthew Shapiro, and Isaac Sorkin. 2019. "Reconsidering the consequences of worker displacements: Firm versus worker perspective." *American Economic Journal: Macroeconomics*, 11(2): 193–227.
- **Gulyas, Andreas, and Krzysztof Pytka.** 2020. "Understanding the Sources of Earnings Losses After Job Displacement: A Machine-Learning Approach." *Working Paper*.
- Jacobson, Louis, Robert LaLonde, and Daniel Sullivan. 1993. "Earnings losses of displaced workers." *The American economic review*, 685–709.
- Lachowska, Marta, Alexandre Mas, and Stephen A Woodbury. 2020. "Sources of displaced workers' long-term earnings losses." *American Economic Review*, 110(10): 3231–66.
- **Lee, David S.** 2009. "Training, wages, and sample selection: Estimating sharp bounds on treatment effects." *The Review of Economic Studies*, 76(3): 1071–1102.
- Leombruni, Roberto, Tiziano Razzolini, and Francesco Serti. 2013. "The pecuniary and non-pecuniary costs of job displacement—The risky job of being back to work." *European Economic Review*, 61: 205–216.
- MMO. 2018. "Ministère du Travail (DARES) [Producteur], Mouvements de Main d'Oeuvre, version 2018 [Fichiers de données], Centre d'Accès Sécurisé aux Données (CASD) [Diffuseur], DOI: 10.34724/CASD.176.3016.V1."
- Nickell, Stephen, Luca Nunziata, and Wolfgang Ochel. 2005. "Unemployment in the OECD since the 1960s. What do we know?" *The Economic Journal*, 115(500): 1–27.

- Schmieder, Johannes F, Till M von Wachter, and Jörg Heining. 2022. "The costs of job displacement over the business cycle and its sources: evidence from Germany." National Bureau of Economic Research.
- **Song, Jae, David Price, Fatih Guvenen, Nicholas Bloom, and Till Von Wachter.** 2019. "Firming up inequality." *The Quarterly journal of economics*, 134(1): 1–50.
- Woodcock, Simon. 2020. "The Effect of the Hartz Labor Market Reforms on Postunemployment Wages, Sorting, and Matching." *IZA Discussion Paper (No. 13300)*.

Online Appendix

The Unequal Consequences of Job Loss

Antoine Bertheau, Edoardo Maria Acabbi, Cristina Barcelo Andreas Gulyas, Stefano Lombardi, Raffaele Saggio

Table of Contents

A	Ado	litional Tables and Figures	23
	A.	Tables	23
	B.	Figures	33
B	San	ple Construction and Analyses	42
	A.	Main Sample	42
	В.	Differences in Observed Characteristics across Countries	42
	C.	The Role of Employers	44
	D.	Employment Coverage	45
	E.	Lee Bounds for Effects on Wages	47
C	Bac	kground: Institutional Settings and Data Sources	48
	A.	Austria	49
	B.	Denmark	50
	C.	France	50
	D.	Italy	52
	E.	Portugal	54
	F.	Spain	55
	G.	Sweden	56
D	Rela	ated Literature	57
	A.	Job Displacement	58

A Additional Tables and Figures

A. Tables

Paper	Country	Year	Tenure	Type of Event	Firm size	Gender	Control group: same employer	Earnings in year 5
Gulyas and Pytka (2020)	Austria	1984-2017	2	Mass layoff $\geq 30\%$	30	Male	No	-16%
Halla, Schmieder and Weber (2020)	Austria	1990-2007	1	Mass layoff	10	Male	No	-20%
Bennett and Ouazad (2019)	Denmark	1990-1994	3	Mass layoff $\ge 30\%$	30	Male	No	- 23%
Roulet (2021)	Denmark	2001-2006	5	Plant closure	5	Both	No	- 12%
Royer (2011)	France	1995-1999	2	Plant closure	10	Both	No	-16%
Brandily, Hémet and Malgouyres (2020)	France	2002-2012	2	Reason for Separation	-	Both	No	-36%
Schmieder, von Wachter and Heining (2022)	Germany	1975-2005	3	Mass layoff \geq 30% Reason for Separation	50	Male	No	-20%
Fackler, Müller and Stegmaier (2021)	Germany	2002-2014	3		-	Male	No	-12%
Leombruni, Razzolini and Serti (2013)	Italy	1989-1994	3	Plant closure	_	Both	No	-9%
Mossucca (2016)	Italy	2005-2010	6	Mass layoff	_	Both	Yes	-9%
Carneiro and Portugal (2006)	Portugal	1991-1998	3	Plant closure	_	Both	Yes	-6%
Raposo, Portugal and Carneiro (2021)	Portugal	1988-2014	2	Plant closure	20	Both	No	-27%
Garcia-Cabo (2018)	Spain	2005-2015	1.5	Reason for separation	-	Both	Yes	-32%
Garda (2012)	Spain	1999-2004	3	Reason for separation	5	Male	No	-25 %
Eliason and Storrie (2006)	Sweden	1987-1988	_	Plant closure	10	Both	No	-11%
Seim (2019)	Sweden	2002-2004	1.5	Reason for separation	5	Male	No	-15%
Jacobson, LaLonde and Sullivan (1993)	USA	1974-1986	6	Mass layoff $\geq 30\%$	50	Both	Yes	-25%
Lachowska, Mas and Woodbury (2020)	USA	2002-2014	6	Mass layoff $\geq 30\%$	50	Both	Yes	-17%

Table A.1: Review of Research Designs and Estimates from the Job Loss Literature

Notes: Selection of papers studying the labor market consequences of job loss in the US and in Europe (the countries in our sample plus Germany). *Year* denotes the years of the displacement. *Tenure* (in years) report the minimum number of years that displaced workers must have worked with their employer up to the moment of displacement; *Firm size* is the minimum firm-size of the employer of displaced workers before displacement; *Type of event* distinguishes how a paper defines a displacement event; *Mass layoff* \geq 30% defines a displacement event when a firm is laying off more than 30% of the its workforce. *Plant-Closure* means that the paper is considering displacement event only when an employer permanently shut-down. *Reason for Separation* means that the paper is using administrative information to determinate the job displacement event. *Control group: same employer* specifies whether the comparison group comprises workers that are restricted to stay with the same employer after the displacement of the treated workers. *Earnings in year 5* reports the job loss effects on earnings 5 years after job displacement in terms of the percent change from the pre-displacement earnings level.

24

	Baselin	ie	Alterna	ative
Denmark				
k = 1	-17.85	(0.155)	-25.47	(0.140)
k = 5	-11.52	(0.196)	-25.29	(0.171)
Sweden				
k = 1	-19.61	(0.240)	-24.73	(0.220)
k = 5	-11.40	(0.329)	-19.83	(0.311)
Austria				
k = 1	-39.00	(0.247)	-45.78	(0.277)
k = 5	-21.96	(0.297)	-36.00	(0.296)
France				
k = 1	-19.56	(0.431)	-25.33	(0.390)
k = 5	-12.25	(0.626)	-25.32	(0.542)
Italy				
k = 1	-39.58	(0.386)	-63.44	(0.496)
k = 5	-27.86	(0.507)	-77.22	(0.537)
Spain				
k = 1	-45.71	(0.551)	-53.71	(0.503)
k = 5	-32.56	(0.672)	-51.21	(0.556)
Portugal				
k = 1	-35.45	(0.286)	-44.49	(0.273)
k = 5	-24.45	(0.332)	-27.38	(0.328)

Table A.2: Earnings Effects with Alternative Control Groups

Notes: Earnings losses 1 and 5 years after job displacement for different definitions of the control group (expressed in proportions with respect to the average pre-displacement labor earnings). *t*^{*} denotes year of job displacement. The *Continuously employed* control group is similar to that in Lachowska, Mas and Woodbury (2020). It is defined by selecting workers who stay employed at the same establishment at which they had at least 3 years of pre-displacement tenure for the entirety of the post-period time window (up to 9 years in total). The *Baseline* control group does not impose the post-displacement restriction and is the control group used in our main analyses.

	Overall gap		Composition part										
		Worker	Employer	Business cycle	Time trend	Total	Total						
	(1)	(2)	(3)	(4)	(5)	(6)	(7)						
Sweden	-0.006	-0.025	0.003	0.000	-0.000	-0.023	0.016						
Austria	-0.099	-0.056	-0.044	0.001	-0.004	-0.104	0.005						
France	-0.006	-0.012	-0.011	0.002	0.002	-0.020	0.014						
Italy	-0.166	-0.063	-0.007	0.003	-0.006	-0.073	-0.092						
Spain	-0.210	-0.023	0.020	-0.000	-0.001	-0.004	-0.205						
Portugal	-0.148	-0.034	-0.003	-0.010	0.001	-0.045	-0.103						

Table A.3: Decomposition of Job Loss Effect on Total Earnings

Notes: Oaxaca-Blinder decompositions by separately comparing each country to Denmark. Column (1) reports the total gap in the job loss effect calculated three years after displacement. Columns (2)-(6) show the part of the gap explained by the following characteristics measured at displacement: worker characteristics (quintiles of worker fixed effects, gender, tenure, age); employer characteristics (quintiles of employer fixed effects, employer size, industry); business cycle conditions (unemployment rate); and timing of separation (quadratic time trend). Column (7) shows the gap part unexplained by the average differences in the observables.

	Italy	Portugal	Spain	France	Austria	Denmark	Sweden
Population of Workers and Firms							
Year of job loss	1993-2016	1992-2017	2007-2019	1994-2016	1987-2018	1983-2017	1994-2016
Individuals: % employees Employers:	6.5	100	4	8	100	100	100
Establishment ID and Firm ID	YES	YES	YES	YES	NO	YES	YES
Public sector employers	YES						
Main Variables							
Earnings include income from							
all jobs	YES	NO	YES	YES	YES	YES	YES
severance payments	NO	NO	NO	YES	NO	YES	YES
self-employment	NO	NO	NO	NO	NO	NO	YES
Days worked	YES	YES	YES	YES	YES	YES	NO
Full time/Part time	YES	YES	YES	YES	NO	YES	YES
Temporary/Permanent contract	YES	YES	YES	YES	NO	NO	NO
Reasons for job separation	YES	NO	YES	NO	NO	NO	NO

Table A.4: Characteristics of Data Sources by Country

Notes: The table summarizes the main characteristics of the datasets. See Appendix C for explanations for each country. *Year of Job loss*: time range of the event-study. % *employees*: the data contains the full population or a sample of X % workers. *Establishment ID and Firm ID*: the data contains both identifiers. *Public sector employer*: the data records *some* jobs in the public sector. See coverage by country in Appendix C. *All jobs*: earnings include all jobs, and not a snapshot in a given month. *Severance payments*: the data contains severance pay or redundancy compensation. *Self-employed*: the data contains labor earnings from non-salaried labor earnings. *Annual days worked*: the data contains the exact number of days covered by an employment contract. *Full time/Part time*: the data contains an indicator to measure whether the job is full-time or part-time. *Temporary/Permanent contract*: the data contains a variable to distinguish between temporary employment contracts and permanent contracts. *Reasons for job separation*: the data contains a variable to identify separation due to a layoff.

	Swe	den	Deni	mark	Aus	stria	Fra	nce	Ita	ly	Spa	ain	Port	ugal
Panel A: Log-daily wag	ze													
Δ in unempl. rate	-0.026	-0.011	-0.007	-0.007	-0.003	-0.005	-0.016	-0.008	-0.060	-0.038	-0.011	-0.006	-0.008	-0.002
	(0.004)	(0.004)	(0.003)	(0.003)	(0.005)	(0.004)	(0.007)	(0.007)	(0.020)	(0.019)	(0.004)	(0.003)	(0.002)	(0.002)
Employer FE	-0.054	0.530	-0.279	-0.263	-0.502	0.048	-0.059	0.212	-0.175	0.056	-0.241	0.094	-0.135	0.139
	(0.033)	(0.033)	(0.007)	(0.008)	(0.015)	(0.015)	(0.022)	(0.023)	(0.011)	(0.011)	(0.022)	(0.022)	(0.013)	(0.012)
Worker FE	0.013	-0.047	-0.332	-0.332	-0.202	-0.337	0.039	-0.061	-0.080	-0.190	-0.092	-0.228	-0.073	-0.178
	(0.009)	(0.009)	(0.005)	(0.005)	(0.008)	(0.008)	(0.012)	(0.012)	(0.009)	(0.009)	(0.019)	(0.017)	(0.006)	(0.006)
Δ in employer FE		1.274 (0.022)		0.015 (0.004)		0.847 (0.010)		0.600 (0.018)		0.583 (0.010)		0.761 (0.020)		0.919 (0.011)
No. of observations	59,583	59,583	80,946	80,946	26,960	26,960	13,241	13,241	25,635	25,635	5,638	5,638	47,311	47,311
Mean dep. var.	-0.057	-0.057	-0.056	-0.056	-0.125	-0.125	-0.047	-0.047	-0.059	-0.059	-0.135	-0.135	-0.034	-0.034
Panel B: Yearly earning	zs													
Δ in unempl. rate	-0.021	-0.008	-0.013	-0.013	0.005	0.003	-0.009	-0.003	-0.102	-0.086	-0.014	-0.010	-0.007	0.000
	(0.003)	(0.003)	(0.003)	(0.003)	(0.007)	(0.007)	(0.008)	(0.008)	(0.027)	(0.026)	(0.005)	(0.005)	(0.006)	(0.006)
Employer FE	0.048	0.535	-0.049	-0.036	-0.604	-0.057	0.084	0.297	-0.149	0.022	-0.268	0.029	-0.090	0.248
	(0.028)	(0.029)	(0.006)	(0.007)	(0.023)	(0.024)	(0.026)	(0.027)	(0.015)	(0.016)	(0.029)	(0.030)	(0.033)	(0.034)
Worker FE	0.057	0.007	-0.062	-0.062	-0.186	-0.321	0.189	0.110	0.001	-0.081	-0.106	-0.226	-0.129	-0.259
	(0.008)	(0.008)	(0.004)	(0.004)	(0.012)	(0.012)	(0.014)	(0.014)	(0.012)	(0.012)	(0.024)	(0.023)	(0.017)	(0.017)
Δ in employer FE		1.062 (0.019)		0.014 (0.004)		0.842 (0.015)		0.473 (0.021)		0.431 (0.014)		0.675 (0.027)		1.134 (0.030)
No. of observations	59,583	59,583	85,218	85,218	26,966	26,966	13,276	13,276	25,635	25,635	5,638	5,638	47,303	47,303
Mean dep. var.	-0.064	-0.064	-0.067	-0.067	-0.145	-0.145	-0.053	-0.053	-0.094	-0.094	-0.191	-0.191	-0.058	-0.058

Table A.5: Cyclicality of Job Loss Effects on Wage and Earnings

Notes: The dependent variable is the wage loss (Panel A) and earnings loss from the pre-displacement level (Panel B) 3 years after job displacement. The baseline controls are: change in unemployment rate, quadratic time trends, firm size, worker's demographics, and employer and worker fixed effects. The individual fixed effects are obtained by subtracting the employer fixed effects from the average pre-displacement wages. The change in the unemployment rate is measured in percentage points. For each country, the second column includes as additional control the change in establishment effect. Section B. discusses the results. Standard errors in parentheses.

	Men		Womer	ı
Denmark				
k = 1	-17.41	(0.191)	-18.86	(0.260)
k = 5	-11.90	(0.243)	-10.56	(0.324)
Sweden				
k = 1	-20.32	(0.290)	-17.85	(0.417)
k = 5	-11.70	(0.402)	-10.62	(0.555)
Austria				
k = 1	-38.58	(0.293)	-39.67	(0.440)
k = 5	-22.64	(0.347)	-20.07	(0.549)
France				
k = 1	-18.77	(0.525)	-21.30	(0.752)
k = 5	-12.16	(0.782)	-12.38	(1.019)
Italy				
$\dot{k} = 1$	-38.32	(0.472)	-42.14	(0.659)
k = 5	-26.99	(0.632)	-29.62	(0.823)
Spain				
k = 1	-44.90	(0.671)	-47.12	(0.954)
k = 5	-31.50	(0.813)	-34.38	(1.178)
Portugal				
k = 1	-34.16	(0.382)	-37.45	(0.417)
k = 5	-24.47	(0.449)	-24.39	(0.473)
Germany				
k = 1	-22.24	(0.19)	-35.49	(0.35)
k = 5	-16.25	(0.42)	-22.35	(0.70)

Table A.6: Effects of Job Loss on Earnings by Country and Gender

Notes: The table shows the coefficients θ_1 and θ_5 from equation (1) for earnings separately estimated by gender within each country. Point-estimates are re-scaled by the average earnings measured in the pre-displacement years and multiplied by 100. Standard errors reported in parentheses and clustered at the individual level. Results from Germany are taken from Schmieder, von Wachter and Heining (2022) and in particular from Figure 1(a), Figure 2(a) (men) and Figure A15(a) and Figure A16(a) (women).

	Ye	ears sin	ce disp	laceme	nt
	k = 1	<i>k</i> = 2	k = 3	k = 4	k = 5
Panel A: All disp	laced we	orkers			
Denmark	0.251	0.181	0.136	0.108	0.089
Sweden	0.186	0.079	0.057	0.042	0.032
Austria	0.244	0.134	0.102	0.083	0.056
France	0.174	0.091	0.069	0.055	0.046
Spain	0.404	0.263	0.205	0.173	0.148
Italy	0.325	0.278	0.253	0.239	0.201
Portugal	0.597	0.478	0.398	0.328	0.236
Panel B: Displace	ed worke	ers who	eventua	lly find	a job
Denmark	0.177	0.100	0.050	0.020	0.000
Sweden	0.159	0.049	0.025	0.010	0.000
Austria	0.197	0.081	0.047	0.026	0.000
France	0.134	0.047	0.023	0.009	0.000
Spain	0.297	0.131	0.062	0.024	0.000
Italy	0.147	0.087	0.056	0.037	0.000
Portugal	0.455	0.294	0.186	0.092	0.000

Table A.7: Share of displaced workers who are still unemployed after displacement

Notes: Share of workers displaced in 2010 who are still non-employed, by years since displacement (*k*). Panel A reports numbers for all displaced workers; in Panel B the displaced workers are conditioned to those who eventually find a job within five years from the displacement event.

Table A.8: Comparing displaced workers who eventually find a job within five years of displacement to those who don't

	Den	mark	Swe	eden	Aus	stria	Fra	nce	Sp	ain	Ita	aly	Port	ugal
Permanently displaced	Yes	No	Yes	No										
Panel A: Worker characte	ristics													
Earnings in $t^* - 2$ Earnings in $t^* - 3$ Age Female Tenure	42,166 39,352 31.82 0.39 5.74	42,711 40,317 33.78 0.37 5.80	33,745 35,665 38.38 0.37 7.58	33,203 32,781 36.58 0.34 7.20	28,433 27,927 40.28 0.45 7.08	30,069 29,435 37.99 0.42 7.32	27,634 26,658 34.58 0.42 7.17	29,932 28,685 37.39 0.36 6.60	23,233 23,280 39.71 0.46 6.47	22,241 22,049 38.01 0.40 6.75	25,318 24,704 39.26 0.47 4.66	23,696 22,635 37.27 0.37 4.66	14,700 15,530 37.13 0.53 11.73	14,571 14,496 35.31 0.46 9.87
Panel B: Employer charac	teristics													
Establishment size Manufacturing Services	376 0.38 0.34	369 0.38 0.34	368 0.43 0.36	384 0.41 0.33	314 0.45 0.18	323 0.47 0.12	262 0.55 0.25	322 0.43 0.34	383 0.27 0.58	337 0.25 0.59	437 0.30 0.47	340 0.38 0.25	335 0.57 0.34	338 0.51 0.40
No. Workers (th.)	5.79	196.12	5.27	103.31	5.03	50.79	1.15	27.48	1.62	13.09	16.18	49.88	45.71	125.43

Notes: This table compares the characteristics of displaced who eventually find a job within five years from displacement (not permanently displaced workers) to those workers who instead are unable to job within five years from displacement (permanently displaced workers).

	Employ- ment effect	Share compliers among employed control workers	Mean wage treated	Mean wage control, excluding compliers	Bound on wage effect
	(1)	(2)	(3)	(4)	(5)
Denmark	-0.036	4%	4.761	4.878	-0.116
Sweden	-0.041	4%	4.407	4.581	-0.174
Austria	-0.091	10%	4.323	4.534	-0.211
France	-0.057	6%	4.342	4.473	-0.131
Spain	-0.156	18%	3.909	4.248	-0.339
Italy	-0.176	23%	4.354	4.616	-0.262
Portugal	-0.173	25%	3.880	4.156	-0.276

Table A.9: Bounds on wag	e losses 5 years	after displacement
--------------------------	------------------	--------------------

Notes: Column 1 displays the employment effects due to displacement observed 5 years following displacement. Column 2 uses the estimates from Column 1 to estimate the share of compliers among the control group, that is, workers that in the presence of displacement would be non-employed five years from displacement. Column 4 then displays the mean wage among control workers assuming that these compliers all belong to the left-tail of the wage distribution observed among control workers. This monotone selection pattern provides therefore an upper bound on the wage effects, computed as the difference between Column 3 and Column 4 and displayed in Column 5, as originally suggested by Lee (2009). See Appendix C for additional details.

B. Figures



Figure A.1: The Effect of Job Loss on Earnings: Evolution for the last 25 years

Notes: The figure shows estimates of earnings losses spanning three decades (1990s-2010s) following job loss as defined in section I. Each plot reports the point estimate – by year of job displacement – of labor earnings losses for the first and the fifth year following involuntary job loss, i.e. θ_1 and θ_5 of the difference-in-difference model (1). Section B. discusses the results.



Figure A.2: Job Loss on Earnings: by Gender

Notes: The figures shows the event-study coefficients θ_k after fitting equation (1) separately by country and gender. The outcome variable is total earnings within the year (rescaled by pre-displacement earnings). Standard errors along with point estimates at k = 1 and k = 5 years from displacement are reported in Table A.6.





Notes: Share of workers by quintiles of wage AKM employer fixed effects measured right before displacement.



Figure A.4: Representativeness of Mass Layoffs Workers for Spain and Italy

Notes: In Spain and Italy, matched employer-employee data provide a reason behind a job separation, see section D. and section F. for details. We use this information to contrast the effects of job displacement on earnings that one would obtain when defining displaced workers to be only those involved in mass-layoff events captured in administrative data— see label "Mass-Layoff Events Only" — to the ones obtained when using the entire set of involuntary separations due to economic reasons that one could measure in the Spanish and Italian context. Specifically, "involuntary separations" due to economic reasons represent all job separations that occurred as a result of the employer facing some economic distress. 95% confidence intervals based on standard errors clustered at the individual level are displayed in the figure.

Figure A.5: Classifying self-employment as non-employment in Sweden



Notes: This figure computes our baseline event-study estimates from equation (1) on employment while setting self-employed jobs recorded in Sweden as non-employment spells. See Appendix D. for details.



Figure A.6: Classifying public sector jobs as non-employment in Denmark, Sweden and France

(b) Classifying as non-employment a selection of public sector jobs not covered in Southern European countries.

Notes: Panel (a) computes our baseline event-study estimates from equation (1) on employment while setting *all* jobs in the public sector Denmark, France and Sweden as non-employment spells. Panel (b) is similar but records as non-employment spells only a subset of public jobs from Denmark, France and Sweden that are particularly unlikely to be recorded in administrative registries from Southern European countries. See Appendix D. for details.



Figure A.7: Explaining Trends in Pay Losses for Italian Displaced Workers

Notes: Each panel shows the displacement effects on either log wage or earnings, 1 or 5 years following the layoff for different cohorts of displaced Italian workers. We overlay to these coefficients the estimates that we obtain on the probability that the first job after displacement is on a temporary job: Finally, we display the results from a simple linear fit for each panel, weighting each square in the scatter-plot by the number of displaced workers observed in a given year. Section D. provides details on the institutional Italian context.



Figure A.8: Permanent withdrawal from the labor market

Notes: The figure reports the event-study coefficients from equation (1). The outcome in the regression is a dummy equal to 1 if worker *i* in period *t* is non-employed in *t* and all the subsequent periods up to year t + 5. Thus, the event-study coefficients captures changes in the probability of displaced workers to permanently withdraw from the labor market relative to their matched control group across countries. The resulting effects displayed in the figure are not necessarily monotonically increasing over time since workers in the comparison group are also allowed to withdraw from the labor market.

B Sample Construction and Analyses

This section provides additional information on the construction of the main sample (Section II) and on the analyses aimed at understanding the heterogeneous job loss effects across countries (Section III).

A. Main Sample

We do not restrict workers from the control group and the treated group to be observed from t^* onwards in order to avoid conditioning on future outcomes. We connect all employment spells at the same establishment in case workers have multiple employment spells during the year.

Treated group. We do not consider workers that find a job in the same firm to be displaced. We control for transitions that follow from change of establishment identifiers due to mergers, split-ups etc. Specifically, we do not allow more than 20 percent of the displaced workers to be reemployed together at the same establishment in the following year. Leaving workers are either non-employed or dispersed to different establishments. Mass-layoff events do not include a "stability" requirement, i.e., employment can increase before or after the drop in mass-layoff event. Treated workers can be treated only once. In order to focus on permanent job separations, treated workers ers who return to their firm up to $t^* + 5$ are dropped from the sample (from both the treated and the comparison group). To avoid classifying mergers and domestic outsourcing as mass layoffs, we create a cross flow matrix of worker flows across firms each year, and exclude mass layoffs from our sample where we observe more than 20 percent of workers jointly moving to another firm.

Control group. Control workers are never treated, to avoid bias induced by control units treated later (see, e.g., de Chaisemartin and D'Haultfoeuille, 2020). However, they are allowed to be coworkers of employees displaced due to a mass layoff, or can be laid off in a given year but not during a mass layoff. Control workers can be used as control only once.

B. Differences in Observed Characteristics across Countries

According to Table 1, the composition of displaced workers differs somewhat across countries. To quantify the role of different observable characteristics of displaced workers in driving the heterogeneous effects of job displacement across countries shown in Figure 1, we perform pairwise Oaxaca–Blinder decompositions between

each country *c* and a reference country *r* (Denmark).¹³ For each country, we regress the individual-level job loss effects on earnings measured in $t^* + 3$ (relative to $t^* - 3$) on a vector of worker- and employer-level characteristics X.¹⁴ Individual-level job loss effects are individual-level difference-in-differences effects computed for each treated-matched control worker pair, see Schmieder, von Wachter and Heining (2022) for a similar approach and Appendix C..2. We use the estimated coefficients from each country-level regression to decompose Δ_c , which denotes the average gap in the job loss effect between country *c* and *r* as follows:

$$\Delta_{c} = \sum_{x \in X} \underbrace{(E[x_{i,c}] - E[x_{i,r}])\beta_{x}^{c}}_{\text{Composition}} + \sum_{x \in X} \underbrace{E[x_{i,r}](\beta_{x}^{c} - \beta_{x}^{r})}_{\text{Unexplained}}$$
(A.1)

The "compositional" part quantifies the differences on the impact of job loss attributable to differences in observables, where the impact of each characteristic x is estimated using the regression coefficients β_x^c of the comparison country. The "unexplained" part quantifies the importance of structural differences between the two countries (unexplained by differences in the observables, which are kept fixed at the reference country's average observed levels).

We focus on the composition part of the cross-country differences by measuring the following characteristics measured right before job displacement: gender, tenure, age, quintiles of worker and employer AKM fixed effects, employer size, economic sector, change in unemployment rate, and quadratic time trends. ¹⁵ These characteristics thus capture potential differences in observable characteristics at the worker and employer level, and the macroeconomic conditions. To facilitate the interpretation of each pairwise comparison, we focus on the job displacement years available for both country *c* and Denmark. Both individual- and worker-level fixed effects are estimated through AKM models and aggregated into country-specific quintiles based on the corresponding AKM sample that excludes displaced workers and their matched control workers.

Table A.3 reports the results of this decomposition exercise.¹⁶ The table shows that compositional differences typically explain only a small part of the total gap in earnings losses between the different countries and Denmark. Figure A.3 provides

¹³We choose Denmark because job loss effects are the smallest there. Choosing Sweden as an alternative reference country yields virtually identical results.

¹⁴We pick three years after job displacement as this permits us to maximize the set of overlapping years where we have data for all the countries without having to rely on very short-run effects.

¹⁵The worker fixed effects are obtained by subtracting the employer fixed effects from the average pre-displacement wages.

¹⁶The sample is restricted until 2015, which is the last year with available information on the OECD indicators of interest (generosity of unemployment benefits, employment protection legislation, active labor market policies, among others).

further visual confirmation of this finding by showing the distribution of the AKM employer-specific wage premium of displaced workers across countries.

C. The Role of Employers

C..1 Sample to estimate employer fixed effects

To limit the extent of noise in the fixed effects estimation, we restrict the samples to workplaces with at least three employees at least once in their histories. Also, to limit the concern that job loss itself contributes directly to the estimates of establishment effect in the AKM model, we exclude treated and control workers from the AKM estimation. Limited mobility of workers across employers can lead to imprecise estimates of establishment fixed effects. This is a first-order concern when performing variance decomposition exercises, which we do not do (see, e.g., Kline, Saggio and Sølvsten, 2020; Bonhomme et al., Forthcoming). We estimate establishment fixed effects for most of the main jobs before and after the relative year of the event t^* .¹⁷

C..2 The cyclicality of wage losses and employer quality

To further investigate the importance of employers in explaining the consequences of job loss, we follow Schmieder, von Wachter and Heining (2022). First, we compute individual-level job loss effects before and after job displacement for each treated–matched control worker pair (between $t^* - 3$ and $t^* + 3$) as follows:.

$$\Delta_{dd}y_{it^*} = (y_{i,T,t^*+3} - y_{i,T,t^*-3}) - (y_{i,C,t^*+3} - y_{i,C,t^*-3}),$$

where $\Delta_{dd} y_{it^*}$ is an estimate of the individual treatment effect from job loss.

Then, we regress the difference in the individual-level job loss effect on the unemployment rate and on additional displaced workers' controls. The set of control characteristics is: female, tenure, age, employer size, quadratic time trends, worker fixed effects, and employer fixed effects. The individual fixed effects are obtained by subtracting the employer fixed effects from the average pre-displacement wages. The impact of the aggregate annual change in the unemployment rate (from $t^* - 1$ to t^*) on wage losses is captured by β . As the mean of the dependent variable, $\Delta_{dd}y_{it^*}$ is negative (ranging from -0.034 in Portugal to -0.135 in Spain, see Table A.5), a negative estimated coefficient indicates that a one percentage point increase in the unemployment rate *increases* wage losses, i.e.:

$$\Delta_{dd} w_{it^*} = \beta \Delta U R_{t^*} + \gamma \hat{\psi}_{J(i,t^*)} + \delta \hat{\alpha}_i + X_i \theta + t^* \pi_1 + t^{*2} \pi_2 + \varepsilon_{it^*}$$
(A.2)

¹⁷For Spain, due to the relatively small sample size, we estimate firm fixed effects.

Table A.5 shows that a positive variation in unemployment rate is strongly associated with the size of wage losses. This result holds for all countries with the exception of Austria, and is in line with Schmieder, von Wachter and Heining (2022) for Germany.¹⁸

Once we include the change in employer fixed effects as additional control, $\Delta_{dd}\hat{\psi}_{J}$, as in model A.3, the conclusion is drastically different.

$$\Delta_{dd}w_{it^*} = \beta \Delta UR_{t^*} + \gamma \hat{\psi}_{J(i,t^*)} + \delta \hat{\alpha}_i + \xi \Delta_{dd} \hat{\psi}_J + X_i \theta + t^* \pi_1 + t^{*2} \pi_2 + \varepsilon_{it^*}$$
(A.3)

In all countries but Austria, the coefficient on the unemployment rate shrinks considerably. And, in most cases, the effect is not statistically significant anymore. This finding clearly indicates that across Europe the reallocation of workers to worse payingemployers in recessions explains the cyclicality of job loss. Even if post-displacement establishment characteristics are endogenous, this correlation, which is empirically verified in many European economies, provides useful information to understand the cyclicality of wages over the business cycle.

D. Employment Coverage

Figure 1 displays remarkably different employment and earnings trajectories across countries following a job displacement event. A potential concern is that this finding is driven by differences in employment coverage across countries. For instance, as displayed in Table A.4, in Sweden self-employed jobs are covered by the data whereas in all the other countries this does not happen to be the case. Yet, this discrepancy does not appear to drive our results. Re-estimating our baseline effects on employment treating self-employed job spells in Sweden as non-employment spells—as it would occur in all the other countries, returns virtually identical results, see Figure A.5.

Another potential source of discrepancy is due to different coverages in public jobs. All our datasets record *some* public jobs. However, coverage of public jobs tends to vary. Denmark, France and Sweden are for instance countries where all public jobs are recorded in their corresponding administrative registers whereas this not occur in countries like Italy, Portugal and Spain where most jobs in public administration (e.g. teachers) are usually not present. To understand the importance of these discrepancies on public jobs for our employment results, we take our data from France, Denmark and Sweden—which cover 100% of all public jobs present in each of these countries—

 $^{^{18}}$ We do not find that variation in the employment rate affects wage losses in Austria. This result can be explained by the fact that there has been little variation in the unemployment rate, ranging from 4% to 6%, in the past 20 years.

and assign all employment spells in public jobs as *de facto* non-employment spells, similarly to what we did above when looking at self-employed jobs in Sweden.

Notice that this clearly represents an extreme or "worst-case" type of scenario as we are implicitly setting *all* public jobs from Denmark, France and Sweden as not-employed, even those type of public jobs that would still instead show up in administrative data from countries like Italy, Portugal or Spain.¹⁹ However, even setting all public jobs as non-employment spells in Denmark, France and Sweden (while maintaining employment spells in the public sector observed in the other countries) does not change the qualitative conclusions. As shown in Figure A.6a, employment effects due to displacement remains about twice as large in Italy, France and Spain compared to what we observe in the remaining countries.

We also produced a more nuanced analysis where we tried to identify public jobs that, to the best of our knowledge, would not be recorded in Italy, Portugal and Spain but that instead would show up in the registers of Denmark, France or Sweden. Treating these jobs as non-employment spells returns virtually identical results as our baseline case displayed in Figure 1, see Figure A.6b.²⁰

All in all, while lack of detailed occupational data from Southern European countries does not permit us to provide a formal cross-walk of public jobs consistently measured across European datasets, the evidence presented in Figures A.6a and A.6b suggests that discrepancies in the coverage of public jobs broadly defined do not appear to drive our employment effects. Dropping public jobs from Denmark, Sweden and France that to the best our knowledge are unlikely to be covered in Italy, Portugal and Spain barely change our point-estimates. Even in a worst case scenario where all public jobs in Denmark, Sweden and France are set as non-employed spells leaves unaltered our main conclusions that displaced workers in Spain, Portugal and Italy are systematically less likely to find a post-displacement job compared to workers from Sweden, France, Austria and Denmark.

These results are consistent with the fact that many public sector jobs in Europe require some form of occupational licensing thus making it difficult for displaced workers – who were all displaced from the private sector – to re-allocate to one of these public jobs (Checchi, Fenizia and Lucifora, 2021). As a result, the potential discrepancies in the coverage of public employment jobs do not appear to drive our baseline employment effects.

¹⁹Employment in companies that are entirely or in part controlled by the national government (e.g. jobs in the national post-office) tend to be present in Southern-European countries.

²⁰For Sweden and Denmark we classify as non-employment the following occupations: teachers, public sector clerks, legislators, and government officials. For France, which adopts an occupation classification not immediately mapped to ISCO codes present in Denmark and Sweden, we set as non-employment the following occupations: administrative and technical executives of the public service; intermediate professions in the public service (administration, security); administrative employees of the public service (service agents and health auxiliaries).

E. Lee Bounds for Effects on Wages

This appendix discusses computation of "Lee Bounds" for the wage effects reported in Figure 1. Let i = 1,...,n index the sample of displaced workers along with their matched counterfactual, control workers. Let $D_i = \mathbf{1}$ {worker *i* is a displaced worker}; $E_{ik} = \mathbf{1}$ {worker *i* is employed *k* years following displacement} and W_{ik} is wages *k* years following displacement. The selection model is assumed to be the following

$$W_{ik} = \alpha + \theta_k D_i + \epsilon_{ik}, \tag{A.4}$$

$$E_{ik}^* = \beta + \gamma_k D_i + v_{ik}, \qquad \gamma_k < 0 \tag{A.5}$$

$$E_{ik} = \mathbf{1} \{ E_{ik}^* > 0 \} \tag{A.6}$$

Among displaced workers, we observe only the "Always Takers" individuals, i.e. those that would have a job in period *k* regardless of whether they were displaced *k* years before. These individuals have $v_{ik} > -\beta - \gamma_k$. However, within the control group, we see both "Always Employed" individuals as well as "Compliers": workers that, if subject to displacement, would actually not be employed in period *k*. Compliers have $-\beta - \gamma_k > v_{ik} > -\beta$. Therefore, the fraction of compliers among employed control workers is given by

$$p_k = \Pr(v_{ik} < -\beta - \gamma_k | v_{ik} > -\beta) \tag{A.7}$$

Under a monotone selection model that assumes that these compliers all belong to the left tail of the wage distribution, we can therefore compute an estimate of the wage effects due to displacement "controlling" for selection following the insights from Lee (2009). First, note that we can identify p_k from

$$p_k = -\frac{E[E_{ik}|D_i = 1] - E[E_{ik}|D_i = 0]}{E[E_{ik}|D_i = 0]}$$
(A.8)

Given an estimate of p_k , one can then assume that all compliers lie in the left part of the distribution of wages, i.e. their wage must be bounded above by w^{p_k} , where w^{p_k} is the p_k centile of the wage distribution among control workers. Therefore, one can construct an estimate of the wage losses due to displacement under a monotone selection model of the type describe above simply by computing:

$$\bar{\mu}_k \equiv E[W_{ik}|E_{ik} = 1, D_i = 1] - E[W_{ik}|E_{ik} = 1, D_i = 0, W_{ik} > w^{p_k}]$$
(A.9)

We stress that this corresponds to a valid estimate of the treatment effect of displacement on wages under the extreme monotone selection model described above, where only highly-productive displaced workers can enter the labor market following the displacement event. In case the latter model does not hold, then $\bar{\mu}_k$ should be viewed as an upper bound on the wage losses due to displacement (with a lower bound that can be constructed simply by computing $E[W_{ik}|E_{ik} = 1, D_i = 1] - E[W_{ik}|E_{ik} = 1, D_i = 0, W_{ik} < w^{p_k})$.

Computation: To compute p_k we simply scale our employment effects from equation (1) at k = 5 by the fraction of control workers employed at k = 5. We use this quantity to construct w^{p_k} and therefore $\bar{\mu}_k$ as displayed in equation (A.9). Results are displayed in Table A.9. The bound is particularly large in Italy, Portugal, and Spain, with a magnitude of almost 30 log points. Therefore, after accounting for re-employment selectivity the cross-country heterogeneity in wage losses are aligned with that found in our main results.

C Background: Institutional Settings and Data Sources

We harmonize the sample construction to make our cross-country variables of interests as comparable as possible. This section reports the details of data sources and key institutional features. In particular, we report the population of firms and workers, and how labor earnings, days worked, and employer size are measured.

Recall that the outcome variables are defined as follows (see Section A.). We define yearly labor earnings, deflated to 2010 EUR, as the sum of labor earnings (possibly from different employers) before taxation. Labor earnings include overtime, bonuses, and severance payments when available. We do not have information on hours worked for all countries. Wages are defined as daily earnings from the main employer, and are computed as labor earnings over days worked. The main employer is the establishment at which annual earnings is the largest. We connect all employment spells at the same establishment in case workers have multiple employment episodes during the year.

The main data sources provided by the distributor in each country used are in the following table. In the replication package we provide all data sources for each country. Below we describe in detail the data used for each country. To control for macroeconomic indicator across countries, we use the hamonized unemployment rate constructed by the OECD (2020).

Country	Databases reference
Austria	A shortene estat de tembre a la (2022)
Austria	Arbeitsmarktuatenbank (2022)
France	CASD (2018)
Denmark	Statistics Denmark (2019)

Italy	INPS (2020)
Portugal	INE (2020)
Spain	Seguridad Social (2022)
Sweden	IFĂU (2018)
Cross- country	OECD (2020)

Notes: Data sources used for each country.

A. Austria

Data Sources We use the administrative records (AMDB) from the social security administration from 1984 through 2019. This data comprises daily information on all jobs and unemployment spells covered by social security (Zweimuller et al., 2009). It contains information on yearly earnings for each worker-establishment pair. The data does not contain information on hours worked. It further contains basic socio-demographic information at the worker level. Each establishment has a unique identifier that allows us to study changes in employer specific characteristics over time. Most public sector employees are subject to social security and are hence in the data. But there exist public sector employees which are not covered by social security, most prominently the police force, teachers, judges, are not covered by the dataset. The self-employed are not reported.

Definition of main variables

- *Employees:* Earnings are the sum of gross labor earnings across all yearly employers.
- *Employers:* The data only contain establishment identifiers, not firm identifiers, hence we cannot delete workers that are considered to be displaced but move to the same firm.

Institutional Settings on Layoffs Employers with more than 20 employees are obliged to notify the Austrian public employment service (AMS) if they intend to collectively dismiss more than a certain number of employees, where the exact threshold depend on firm size. Furthermore, firms and work councils must agree on a social plan, which can include voluntary severance payments, financial interim aid, reimbursement of costs for education, training or job interviews. Until 2002, long-tenured workers were eligible for severance pay. The Employees Income Provision Act in 2003 eliminated severance pay and replaced it with monthly employer contributions into pension accounts accessible during unemployment spells. See Andreas Kettemann, Francis Kramarz and Josef Zweimüller (2017).

Related studies Gulyas and Pytka (2020) is the closest paper. They use a recent machine learning method to uncover the sources behind job loss. They find that the main sources behind job losses are related to employer specific factors (AKM firm's wage premiums and the availability of well paying jobs in the local labor market).

B. Denmark

Data Sources Our main data source is the IDA dataset from 1980 to 2018, provided by Denmark Statistics (IDAN, IDAS, IDAP). IDA contains the universe of Danish residents with establishment and firm identifiers. There is no information on job separations, nor on contract type (temporary or permanent). The data source changed in 2008, which impacts the computation of the days worked and labor earnings variables. We also use the dataset IND for information on earnings.

Definition of main variables

- *Employees:* Earnings comprise all salary-related income in a year.
- *Employers:* The number of employees in the establishment on November 28th is used as establishment size. Industry group follows the NACE classification. Public sector employers include the state and municipalities.

Institutional Setting Employers have to inform the local authorities and start negotiating with a worker representative in cases of mass layoffs.

Notice periods and severance payments vary from one to six months, depending on workers ´ tenure. In the event of large mass layoffs, special funding (*Varslingspulje*) is granted to local job centers. The OECD (2016*a*) and the European Restructuring Monitor website provide further explanations of the institutional setting.

Unemployment insurance is voluntary. Low-income members of the insurance system receive benefits worth 90% of their pre-unemployment salary, but the replacement rate is lower for middle and top income groups. For an average production worker, the replacement rate is less than 50% (see Andersen et al. (2020)). A string of reforms changed labor market policies in the mid-1990s (see Andersen and Svarer (2007)).

Related Studies Roulet (2021) finds a similar impact of job displacement using plant closure as the displacement event. In contrast, Bennett and Ouazad (2019) find larger impacts.

C. France

Data Source We use the DADS (DADS-Panel, 2018) dataset that includes a sample of salaried workers from 1991 to 2018. The dataset is provided by the CASD. ²¹ Until 2001, the sample corresponds to a 1/25 random sample. Starting in 2002 the sample was doubled. The dataset contains establishment and firm identifiers, and records public sector jobs. The panel does not follow workers outside salaried jobs (e.g., self-employed workers).²² Furtermore, we also use the datasets DADS Poste (DADS-Poste, 2018), DADS Etablissements (DADS-Etablissement, 2015), and the dataset MMO (MMO, 2018).

Definition of main variables

- *Employees:* Earnings include all payments to workers; profit-sharing schemes, employee savings schemes, severance payments and perks.
- *Employers:* The number of employees in the establishment on December 31st is the establishment size. Industry classification is based on a 5 group economic activity category.

Institutional Setting A plan that aims to reduce the numbers of layoffs is mandatory in firms with more than 50 employees, in which at least 10 employees will be laid off within 30 days. Legal severance pay comes to approximately 25% of the monthly reference wage. Severance payments can explain the increase of daily wages in t=0 reported in Figure 1.

Unemployment benefits end after 24 months for workers below 50 years old, and the net (and constant over the unemployment spell) replacement rate is 71%. Special benefits are granted to displaced workers. The replacement rate can be up to 100% of the previous net salary for one year, with special counselling and training. The French labor market has become segmented over the last three decades, with an increase of

²¹We use the dataset "DADS panel tous salariés 2018", starting in 1976. This dataset is constructed by the French national statistical office (INSEE) from social security records 1) for private sector that establishments must filled once a year for each employee (*DADS*, *Déclarations Administratives de Données Sociales*) and 2) for central government public employees (*FPE*, *fichiers de paye des agents de l'état*). We start our sample in 1991 for several reasons. The years 1981, 1983 and 1990 are not available. Public servants were gradually recorded in the 1980s. Hospital public servants (fonctionnaires de la fonction publique hospitalière) are integated in 1984.Regional public servants (fonctionnaires des collectivités territoriales) and state level public servants (fonctionnaires de la fonction publique d'État) are integrated in 1988. Earnings, is estimated before 1993. The quality of the worker identifier is poorer in the 1980s (Kramarz and Perez-Duarte, 2009). In the 1980s, INSEE constructed the DADS dataset from paper forms, and not from electronic medium . (tape of diskette).

²²We are grateful to Pauline Carry, Bérengère Patault, and Elio Nimier-David for their help on the French data.

jobs under fixed-term contracts. Moreover, part-time unemployment (*Activité Réduite*) is increasingly used (Benghalem, Cahuc and Villedieu, 2021).

Related Studies Royer (2011), Frocrain (2018) and Brandily, Hémet and Malgouyres (2020) evaluate the impact of establishment closures on workers. Brandily, Hémet and Malgouyres (2020) identify job losses from two samples: 1. workers that receive unemployment insurance as "laid-off for economic reasons" and 2. workers employed in establishments that close. They document a long term reduction of 36% of earnings (≈ 15 % in sample 2.) and 11% of hourly wages (≈ 6 % in sample 2.). The firm (AKM) wage premium explains 84.5% (sample 1) and 95.5% (sample 2) of the long-term hourly wage losses.

D. Italy

Data Sources The main data source is derived from social security records stored by the Social Security Institute (Istituto Nazionale Previdenza Sociale, INPS). This dataset, which we label INPS-LOSAI, contains roughly 6.5% of the universe of workers present in the universe of INPS records. The panel records all employment spells in private-sector salaried-jobs. However, jobs under employers that are controlled in part or entirely by the government (e.g. job in the national post-office) tend to also be included. Attrition can be due to unemployment, self-employment or employment in public administration (e.g. teacher). Information on whether a job is under temporary contract and the reasons behind a job termination is available since 1998 and 2005, respectively.

Definition of main variables

- *Employees*: Earnings includes base labor earnings, regular benefits (based on seniority) and irregular benefits (e.g., profit distributions, premiums at the firm level, holiday bonuses are also included). Earnings are top coded at roughly the 99.5 percentile (Hoffmann, Malacrino and Pistaferri, 2021).
- *Employers*: Yearly information on employer size is collected within the LOSAI dataset in various bins (0-5, 6-10, 11-15, 16-20, 21-25, 26-30, 31-40, 41-50, 51-100, 101-200, 201-300, 301-400, 401-500, 500). We take the midpoint in each bin and define that as the employer size for a given year. An employer is defined based on the employer identifier provided by INPS. As in Spain (see below), we consider an employer to be involved in a mass layoff when one of these two situations occurs: (i) when the employer experiences a reduction in the number of workers employed of more than 30% relative to the previous year or (ii) the

reason of job separation given to INPS by the employer is "firing for economic reasons" which represents scenarios in which the employer is laying off part or all of its workforce because of financial difficulties. Below we show that similar earnings losses are obtained using only (i).

Institutional settings Employment legislation surrounding layoffs typically applies to firms that have more than 15 employees (Kugler and Pica, 2008). Sectoral bargaining agreements might provide specific criteria on which workers should be subject to the layoff. Prior to the layoff, it is typical to observe some workers receiving zero hours contracts (Giupponi and Landais, 2020). Following the layoff, the worker receives the so-called "trattamento di fine rapporto (TFR)" which is calculated as a full year of salary divided by 13.5 plus approximately 1.5% for each year of tenure.

Related Studies The closest paper to our study using Italian data is Mossucca (2016). She estimated job displacement effects using INPS data. However, she does not have information on firm-size and, therefore, uses worker-level information on whether workers were assigned to zero-hours contracts to proxy for mass-layoff events.

Downward Trend in Pay Losses during the 2000s Earnings losses for Italian displaced workers appear to experience a downward trend during the 2000s.

It is worth investigating the causes of this particular downward trend. The decade 2000-2010 is a period of profound transformations of the Italian labor market. The landmark of this process of transformation is the dualization of employment contracts. Temporary employment contracts were liberalized during this period. This liberalization was achieved while maintaining rigid levels of employment protection for permanent contract workers (Boeri, 2011; Daruich, Addario and Saggio, 2020).

This leads to the question: Are Italian workers who were displaced in the 2000s experiencing larger earnings losses because they are more likely to have a temporary job following a job loss? Figure A.7 overlays the event study coefficients on earnings and wage losses experienced by workers displaced in different years with the event study of the probability that the first job obtained after a layoff is on a temporary employment contract. It appears that the effect on the share of displaced workers obtaining a temporary job following displacement predicts wage and earnings losses well, both in the short and long-run. The negative association between earnings losses and temp-share following displacement. Instead, a substitution effect appears to dominate: displaced workers are increasingly more likely to obtain a temporary job (as opposed to a permanent one) and this causes significant wage and earnings losses both in the short and in the longer run.

In conclusion: the downward trajectory in pay losses appears to be due in part to by changes to Italian institutions that facilitated the hiring of workers on a temporary basis. This finding echoes the ones in Woodcock (2020) who found that German workers displaced after the passage of the so called Hartz-reforms experienced (i) larger wage losses (ii) a substantial part of these wage losses is due to workers increasingly sorting into temporary jobs.

E. Portugal

Data Sources The main data source is the *Quadros de Pessoal* (hereafter QP) for the 1987-2018 period.²³ The data are gathered annually by the Portuguese Ministry of Employment through an questionnaire that every establishment is obliged by law to fill in. The dataset does not cover the public administration and non-market services, whereas it covers partially or fully state-owned firms. The dataset covers virtually the entire population of firms. The dataset contains a snapshot of firms' employment each year (the reference month being March before 1994 and October from then on). It contains information on industry (NACE 2), hiring date, the kind of job contract (fixed-term or open-ended), the effective number of hours worked, and different types of compensation. This implies that jobs (hence earnings, days worked and daily wages) are not recorded for a worker who is not employed in October.

Finally, due to the fact that the year 2001 is missing from the QP at worker level, we exclude the years 2000 and 2001 as possible treatment years. We also remove from the treatment years the year 1999, due to the disproportionate and implausible amount of displaced workers who disappear from the dataset compared to other years, which makes the year a clear outlier. See Acabbi, Panetti and Sforza (2021) for additional details about the data source.

Definition of main variables

- *Employees:* Earnings include base earnings, regular benefits (based on seniority) and irregular benefits (profit distributions and premiums). Earnings do not contain severance payments.
- *Employers:* Number of employees in establishments are measured at the end of October. The definition of a mass layoff is based on the variation in employment from October to October each year.

²³We are grateful to Pedro Raposo for his help to access to the data. The INE project number associated with our data request is PED-452199310.

Related studies The closest paper to our study using Portuguese data is Pedro Raposo, Pedro Portugal and Anabela Carneiro (2021). They evaluate the sources of wage losses of workers displaced from 1988 to 2011, with different sample restrictions.

They find that sorting into lower paying job titles represents the largest component of the monthly wage loss of displaced workers, accounting for 37% of the total average monthly wage loss compared to 31% for the firm and 32% for the match effects.

F. Spain

Date Sources We use administrative data from the Continuous Sample of Working Histories (*Muestra Continua de Vidas Laborales*, MCVL) for the period 2005-2019, provided by the Spanish Social Security Administration. This sample is a 4% random draw from the universe of Social Security records, employed and unemployed workers and retired people in the reference year. This sample also offers retrospective information of the entire labor history of workers. Around one third of the public sector employees are not included in the sample (excluded from the General Regime of the Social Security).

The dataset contains monthly information on the number of days worked, the kind of job contract (open-ended or fixed-term) and the working time (whether full-time or part-time job, and the fraction of working time) for all employers. Hours worked are not available.

Definition of main variables

• *Employees*: Earnings refer to the monthly contribution to Social Security that can be top- and bottom-coded, including annual bonuses and excluding overtime hours and severance payments. The minimum and maximum limits vary by workers and over time, depending on the minimum wage and inflation. The data also provides information on total yearly earnings (i.e., not top and bottom coded) coming from tax records.

As a robustness check, we have reestimated the consequences of job loss in earnings and wages to assess that the results are almost statistically identical when we use information on total taxable labor earnings for the period with both income sources available. They only differ significantly in the pre-displacement year ($t^* - 1$) and in the year of mass layoff (t^*) as earnings from tax records include severance payments.

• *Employers*: The number of employees in an establishment is available for the month of April one year later.

Hence, we redefine our reference year in the analysis from May to April of next year. This makes the yearly information on the number of employees in the establishment coincide with the end of the reference year (for instance, year 2018 in our analysis covers from May of the calendar year 2017 to April of 2018).

An employer is involved in a mass layoff when one of these two situations occur: (i) the reason for job separation given to Social Security by firms is a permanent collective dismissal (*Expediente de Regulación de Empleo*, ERE) or (ii) when the establishment experiences a reduction in the number of workers employed in more than 30% with respect to the previous year. Figure A.4 shows that estimates of earnings losses are similar with or without using the condition (i) (*ERE*).

Institutional Setting Firms must ask for authorization for a collective dismissal when the number of dismissed workers exceeds a certain threshold in a three-month period depending on the initial firm size (*Expediente de Regulación de Empleo*, ERE). In collective dismissals, the legal severance payments are the salaries of 20 days per year worked with a maximum level equal to 12 months earnings. In cases of unfair dismissals of permanent workers, severance payments are the earnings of 33 days per year worked with a maximum payment of 24 months. In cases of fixed-term contracts, it is 8 days per year worked and 12 days since 2015 (see Barceló and Villanueva (2016)). The maximum duration of unemployment benefits is 24 months. The replacement rate of unemployment benefits is 70% of the contribution base in the first 6 months and 50% afterwards. The amount of unemployment benefits varies

between $527.24 \in$ and $1,482.86 \in$ in 2019. The use of fixed-term contracts is very high in Spain. Since 2015, the maximum length of a short-term contract is three years, which can be extended one year more in some cases.

Related studies Garda (2012) finds wage drops in the long run of roughly 10% for permanent contract and 5% for fixed term contract. Garcia-Cabo (2018) also finds wage losses of 15% on average, but the sample restriction is different.

G. Sweden

Data Sources We use the RAMS matched employer–employee database from Statistics Sweden (SCB). The database contains full population-level information on the gross labor earnings paid for each employment spell (public and private sector jobs). RAMS does not provide information on the reason for layoffs nor on the nature of the contract. We complement the employment information with socioeconomic characteristics from the LOUISE dataset (SCB). RAMS is also used to compute firm size and employer in November.

Definition of main variables

• *Employees:* Earnings is the sum of gross labor earnings across all employers. The employment spells are used to compute the number of days employed at the primary employer (by multiplying the corresponding number of months worked by 30) and the daily earnings at the primary employer.

Institutional setting The Swedish institutional setting is similar in many respects to that of Denmark and other Nordic countries when it comes to unemployment insurance and active labor market programs. The Swedish model integrates flexibility for employers and security for employees. Workers can *voluntarily* insure against job loss, which gives them eligibility to receive unemployment benefits.

The unemployment insurance system is characterized by conditionality: unemployment benefits can be subject to suspension if jobseekers do not fulfill the job search requirements (see Lombardi 2019).

Job security councils help workers who lose their jobs during mass layoffs to transition towards a new job. The transition services provided include training and start-up support to employees.

One specificity of the Swedish system in the case of mass layoffs is a set of rules that go under the name LIFO ("last-in-first-out"; see OECD, 2016*b*).

This implies that workers with lower tenure leave the firm first, whereas longertenured workers are laid off at a lower priority. In practice, firm-level bargaining can imply deviations from LIFO rules. OECD (2016*b*) gives an overview of the institutional setting.

Permanent contracts are the main rule. Fixed-term employment contracts must be provided by law or collective agreement.

Related studies Eliason and Storrie (2006) study long-term effects of job displacements in 1987 up to 12 years later. The lack of post-displacement earnings recovery is attributed to the 1990s Swedish financial crisis. Seim (2019) studies displacement effects in Sweden for displacements in 2002–2004 by using information that allows resignations to be distinguished from actual displacements.

Five years after displacement, our earnings loss effects are similar to those in Seim (2019), both in levels (around 4,000 Euros in 2010 currency) and as percentage change from the pre-displacement level (about 10% losses).

Cederlöf (2021) provides job loss estimates using a mass layoff design similar to the one we implement.

D Related Literature

This section reviews recent theoretical frameworks and empirical work. See Carrington and Fallick (2017) for a review.

A. Job Displacement

A..1 Theoretical framework

Key ideas. Some models are based on *loss of skills*. Loss of skills can be split into two categories. First, firm-specific skills are acquired over time during the employment spell and are mainly valuable in the current job (Becker, 1964; Lazear, 2009). Second, general skills can be lost over the unemployment period (Pissarides, 1992; Ljungqvist and Sargent, 1998). In the class of search models, *losses in firm rents or match components* explain earnings losses. Over the employment spell in search models, wages rise with tenure as wages are renegotiated (Cahuc, Postel-Vinay and Robin, 2006), or simply through commitment (Burdett and Coles, 2003).

In job matching models, such as in Jovanovic (1979), workers lose a fixed component of their wage which is specific to a match. Recent models combine some of those mechanisms. For instance, see Krolikowski (2017), Jarosch (2021), Huckfeldt (2022), Kenneth Burdett, Carlos Carrillo-Tudela and Melvyn Coles (2020), Acabbi, Alati and Mazzone (2022).

A..2 Empirical Evidence

US evidence. Davis and Von Watcher (2011) report a range of earnings losses going from -18% to -25% depending on displacement years (see Hall (2011) for a discussion). Lachowska, Mas and Woodbury (2020) study displacement events from 2008-2010 for Washington State. They find a reduction of 15% in earnings, 2.7% in hours worked and 4.9% in hourly wages up to five years after the event. Match effects, as in Woodcock (2015), explain 57% of the job loss, while AKM firm fixed effects explain 17%. In their sample, the AKM firm fixed effect is not important as 70% of workers move to a better or same AKM quintile firm. Using Ohio data, Moore and Scott-Clayton (2019) report that between 16% to 24% of long-run earnings losses is explained by firm rents.

European evidence. Schmieder, von Wachter and Heining (2022) study job displacement in 1980-2009 in Germany and find a 10% decrease in earnings up to 10 years after displacement. In contrast to evidence based on U.S data, they conclude that a large part of wage losses and a substantial degree of their cyclicality can be explained by the reduction of average wage levels of new employers. Schmieder, von Wachter

and Heining (2022) find that, going from peak to trough of the business cycle in Germany raises short-term earnings losses from -13% to -25%, similar in magnitude to Davis and Von Watcher (2011). Fackler, Müller and Stegmaier (2021) shows that wage losses for plant closures in Germany depend on pre-displacement plant size.

Raposo, Portugal and Carneiro (2021) study job loss in 1988-2014 in Portugal. In their sample, 46% of the wage loss is due to sorting into lower paying jobs, 27% of the loss due to match effects, and the remaining 27% is accounted for the drop in employer fixed effects. OECD (2018) reports earnings losses using a mix of survey and administrative data over the period 2000 to 2005 for several OECD countries.

Comparing existing evidence. It is not possible to compare the above-mentioned results because they apply different econometric models and impose different sample restrictions. This point is illustrated in Table A.1. In terms of methods, Raposo, Portugal and Carneiro (2021) estimate an AKM model, but include job titles, that blend skill requirements of the worker and the bargaining power of the workers' organizations. Sample selection also greatly differs across the papers mentioned above. For instance, the set of comparison workers are different across studies. Schmieder, von Wachter and Heining (2022) build their control as workers that do not leave the firm up to t = 0, while Lachowska, Mas and Woodbury (2020) restrict to at least t = 4. Previous research shows that different comparison groups lead to different earnings losses (Krolikowski, 2018; Cederlöf, 2021).

References

- Acabbi, Edoardo, Ettore Panetti, and Alessandro Sforza. 2021. "The Financial Channels of Labor Rigidities: Evidence from Portugal." *Working Paper*.
- Acabbi, Edoardo Maria, Andrea Alati, and Luca Mazzone. 2022. "A Labor Market Sorting Model of Hysteresis and Scarring." *Available at SSRN*.
- Andersen, Asger, Amalie Jensen, Niels Johannesen, Claus Kreiner, Søren Leth-Petersen, and Adam Sheridan. 2020. "How Do Households Respond to Job Loss? Lessons from Multiple High-Frequency Data Sets."
- Andersen, Torben, and Michael Svarer. 2007. "Flexicurity: labour market performance in Denmark." *CESifo Economic Studies*, 53(3): 389–429.
- Arbeitsmarktdatenbank. 2022. "Wages (hv_bmg), social security (hv_epi_stamm), workers information (hv_pn), establishments (hv_dg_konto)." https: //arbeitsmarktdatenbank.at/index_start.htm?userName=AMDBAMS1 (Accessed: 2022).
- **Barceló, Cristina, and Ernesto Villanueva.** 2016. "The response of household wealth to the risk of job loss: Evidence from differences in severance payments." *Labour Economics*, 39: 35–54.
- Becker, Gary. 1964. Human Capital.
- **Benghalem, Hélène, Pierre Cahuc, and Pierre Villedieu.** 2021. "The Lock-in Effects of Part-time Unemployment Benefits." *Working Paper*.
- **Bennett, Patrick, and Amine Ouazad.** 2019. "Job displacement, unemployment, and crime: Evidence from danish microdata and reforms." *Journal of the European Economic Association*.
- **Boeri, Tito.** 2011. "Institutional Reforms and Dualism in European Labor Markets." *Handbook of Labor Economics*, 4: 1173–1236.
- Bonhomme, Stéphane, Kerstin Holzheu, Thibaut Lamadon, Elena Manresa, Magne Mogstad, and Bradley Setzler. Forthcoming. "How Much Should we Trust Estimates of Firm Effects and Worker Sorting?" *Journal of Labor Economics*.
- **Brandily, Paul, Camille Hémet, and Clément Malgouyres.** 2020. "Understanding the Reallocation of Displaced Workers to Firms." *PSE Working Paper 2020-81*.
- Burdett, Ken, and Melvyn Coles. 2003. "Equilibrium wage-tenure contracts." *Econometrica*, 71(5): 1377–1404.

- **Burdett, Kenneth, Carlos Carrillo-Tudela, and Melvyn Coles.** 2020. "The cost of job loss." *The Review of Economic Studies*.
- **Cahuc, Pierre, Fabien Postel-Vinay, and Jean-Marc Robin.** 2006. "Wage bargaining with on-the-job search: Theory and evidence." *Econometrica*, 74(2): 323–364.
- **Carneiro, Anabela, and Pedro Portugal.** 2006. "Earnings losses of displaced workers: Evidence from a matched employer-employee data set."
- **Carrington, William, and Bruce Fallick.** 2017. "Why do earnings fall with job displacement?" *Industrial Relations: A Journal of Economy and Society*, 56(4): 688–722.
- CASD. 2018. "Insee [Producteur], Panel tous salariés 2018 [Fichiers de données], and DADS Etablissements, DADS Poste and Mouvements de Main-d'Oeuvre (MMO) via Centre d'Accès Sécurisé aux Données (CASD) [Diffuseur],." https://www.casd. eu/en/contacts-title (Accessed: 2019).
- **Cederlöf, Jonas.** 2021. "Reconsidering the Cost of Job Loss: Evidence from Redundancies and Mass Layoffs."
- **Checchi, Daniele, Alessandra Fenizia, and Claudio Lucifora.** 2021. "Public sector jobs: Working in the public sector in Europe and the US."
- **Daruich, Diego, Sabrina Di Addario, and Raffaele Saggio.** 2020. "The Effects of Partial Employment Protection Reforms: Evidence from Italy."
- **Davis, Steven, and Till Von Watcher.** 2011. "Recessions and the Costs of job Loss." *Brookings Papers on Economic Activity: Fall* 2011, 1.
- de Chaisemartin, Clément, and Xavier D'Haultfoeuille. 2020. "Two-way fixed effects estimators with heterogeneous treatment effects." *American Economic Review*, 110(9): 2964–96.
- Eliason, Marcus, and Donald Storrie. 2006. "Lasting or latent scars? Swedish evidence on the long-term effects of job displacement." *Journal of Labor Economics*, 24(4): 831–856.
- **Fackler, Daniel, Steffen Müller, and Jens Stegmaier.** 2021. "Explaining wage losses after job displacement: Employer size and lost firm rents." *Journal of the European Economic Association*.
- **Frocrain, Philippe.** 2018. "Globalization and the labor market: A study of tradable and non-tradable jobs in France." PhD diss. PSL Research University.

- **Garcia-Cabo, Joaquin.** 2018. "The Macroeconomic Effects of Employment Protection on Human Capital and Jobs."
- **Garda, Paula.** 2012. "Wage losses after displacement in Spain. The role of specific human capital." *Universitat Pompeu Fabra (mimeo)*.
- **Giupponi, Giulia, and Camille Landais.** 2020. "Subsidizing Labor Hoarding in Recessions: The Employment & Welfare Effects of Short Time Work." *CEPR Discussion Paper*.
- **Gulyas, Andreas, and Krzysztof Pytka.** 2020. "Understanding the Sources of Earnings Losses After Job Displacement: A Machine-Learning Approach." *Working Paper*.
- Halla, Martin, Julia Schmieder, and Andrea Weber. 2020. "Job displacement, family dynamics, and spousal labor supply." *American Economic Journal: Applied Economics*, 12(4): 253–87.
- **Hall, Robert.** 2011. "Comments and Discussion on "*Recessions and the Costs of Job Loss*" by Steven Davis and Till Von Wachter." *Brookings Papers on Economic Activity*, 56–60.
- **Hoffmann, Eran, Davide Malacrino, and Luigi Pistaferri.** 2021. "Labor Market Reforms and Earnings Dynamics: the Italian Case." *IMF Working Paper*.
- **Huckfeldt, Christopher.** 2022. "Understanding the scarring effect of recessions." *American Economic Review*, 112(4).
- IFAU. 2018. "Employers (RAMS), demographics (Louise), wages (Structure of earnings)." https://www.ifau.se/en/About-IFAU/contact (Accessed: 2020).
- INE. 2020. "Firms (QP_Emp), establishments (QP_Est), workers (QP_Trabalhadores)." https://www.ine.pt/xportal/xmain?xpid=INE&xpgid=ine_main, then press Sobre o INE>O que fazemos>Informação Estatística>Investigadoras/es>Bases de dados>6.3 Quadros de Pessoal, element 6.3, (Accessed: 2020).
- INPS. 2020. "INPS-LoSAI dataset. Workers' spells (Rapporti_lavoro1985_2018), demographics (ANAGRAFICA_2018), employers (IMPRESE_1990_2014, MATR_LOSAI)." http://dati.lavoro.gov.it/microdati-la-ricerca (Accessed: 2020).
- Jacobson, Louis, Robert LaLonde, and Daniel Sullivan. 1993. "Earnings losses of displaced workers." *The American economic review*, 685–709.
- Jarosch, Gregor. 2021. "Searching for job security and the consequences of job loss." *NBER Working Paper*.

- Jovanovic, Boyan. 1979. "Job matching and the theory of turnover." *Journal of political economy*, 87(5, Part 1): 972–990.
- **Kettemann, Andreas, Francis Kramarz, and Josef Zweimüller.** 2017. "Job mobility and creative destruction: flexicurity in the land of Schumpeter."
- Kline, Patrick, Raffaele Saggio, and Mikkel Sølvsten. 2020. "Leave-out estimation of variance components." *Econometrica*.
- Kramarz, Francis, and Sébastien Perez-Duarte. 2009. "Wage structure in France, 1977-96." In *The Structure of Wages: An International Comparison*. 401–418. University of Chicago Press.
- **Krolikowski, Pawel.** 2017. "Job ladders and earnings of displaced workers." *American Economic Journal: Macroeconomics*, 9(2): 1–31.
- **Krolikowski, Pawel.** 2018. "Choosing a Control Group for Displaced Workers." *ILR Review*, 71(5): 1232–1254.
- **Kugler, Adriana, and Giovanni Pica.** 2008. "Effects of employment protection on worker and job flows: Evidence from the 1990 Italian reform." *Labour Economics*, 15(1): 78–95.
- Lachowska, Marta, Alexandre Mas, and Stephen A Woodbury. 2020. "Sources of displaced workers' long-term earnings losses." *American Economic Review*, 110(10): 3231–66.
- Lazear, Edward. 2009. "Firm-specific human capital: A skill-weights approach." *Journal of Political Economy*, 117(5): 914–940.
- **Lee, David S.** 2009. "Training, wages, and sample selection: Estimating sharp bounds on treatment effects." *The Review of Economic Studies*, 76(3): 1071–1102.
- Leombruni, Roberto, Tiziano Razzolini, and Francesco Serti. 2013. "The pecuniary and non-pecuniary costs of job displacement—The risky job of being back to work." *European Economic Review*, 61: 205–216.
- Ljungqvist, Lars, and Thomas Sargent. 1998. "The European unemployment dilemma." *Journal of political Economy*, 106(3): 514–550.
- **Lombardi, Stefano.** 2019. "Threat Effects of Monitoring and Unemployment Insurance Sanctions: Evidence from Two Reforms." IFAU working paper, 2019:22.
- Moore, Brendan, and Judith Scott-Clayton. 2019. "The Firm's Role in Displaced Workers' Earnings Losses." *NBER Working Paper*.

- **Mossucca, Rossella.** 2016. "Job Displacement: Comparative Evidence for Italy." *Available at SSRN 3012142.*
- **OECD.** 2016*a*. "OECD Back to Work Denmark: improving the re-employment prospects of displaced workers."
- **OECD.** 2016*b*. "OECD Back to Work Sweden: improving the re-employment prospects of displaced workers."
- **OECD.** 2018. "Back to work: Lessons from nine country case studies of policies to assist displaced workers." *OECD Employment Outlook (Chapter 4)*.
- **OECD.** 2020. "Unemployment rate (indicator)." https://doi.org/10.1787/ 52570002-en (Accessed: 2020).
- **Pissarides, Christopher.** 1992. "Loss of skill during unemployment and the persistence of employment shocks." *The Quarterly Journal of Economics*, 107(4): 1371–1391.
- **Raposo, Pedro, Pedro Portugal, and Anabela Carneiro.** 2021. "The sources of the wage losses of displaced workers: the role of the reallocation of workers into firms, matches, and job titles." *Journal of Human Resources*, 786–820.
- Roulet, Alexandra. 2021. "The Causal Effect of Job Loss on Health: The Danish Miracle?"
- **Royer, Jean-François.** 2011. "Évaluation des effets des brusques fermetures d'établissements sur les trajectoires salariales." *Economie et Statistique*.
- Schmieder, Johannes F, Till M von Wachter, and Jörg Heining. 2022. "The costs of job displacement over the business cycle and its sources: evidence from Germany." National Bureau of Economic Research.
- Seguridad Social. 2022. "Muestra Continua de Vidas Laborales, MCVL (2005-2019)." https://www.seg-social.es/wps/portal/wss/internet/ EstadisticasPresupuestosEstudios/Estadisticas/EST211 (Accessed: 2022).
- **Seim, David.** 2019. "On the incidence and effects of job displacement: Evidence from Sweden." *Labour Economics*, 57: 131–145.
- Statistics Denmark. 2019. "IDA datasets (IDAN, IDAS and IDAP), and individual income dataset (IND)." https://www.dst.dk/en/TilSalg/Forskningsservice (Accessed: 2020).
- Woodcock, Simon. 2015. "Match effects." Research in Economics, 69(1): 100–121.

- Woodcock, Simon. 2020. "The Effect of the Hartz Labor Market Reforms on Postunemployment Wages, Sorting, and Matching." *IZA Discussion Paper (No. 13300)*.
- Zweimuller, Josef, Rudolf Winter-Ebmer, Rafael Lalive, Andreas Kuhn, Jean-Philippe Wuellrich, Oliver Ruf, and Simon Buchi. 2009. "Austrian social security database." *Available at SSRN 1399350*.